

Psychosocial and psychological interventions for preventing postpartum depression (Review)

Dennis CL, Creedy DK



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2008, Issue 4

http://www.thecochranelibrary.com



TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
METHODS	3
RESULTS	5
DISCUSSION	9
AUTHORS' CONCLUSIONS	10
ACKNOWLEDGEMENTS	11
REFERENCES	11
CHARACTERISTICS OF STUDIES	16
DATA AND ANALYSES	29
Analysis 1.1. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 1 Depressive	
symptomatology at final study assessment (variously defined).	35
Analysis 1.2. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 2 Depressive	
symptomatology at final study assessment (EPDS > 12)	37
Analysis 1.3. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 3 Mean depression	
scores at final study assessment.	38
Analysis 1.4. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 4 Depressive	
symptomatology at 8, 16, 24 weeks postpartum (variously defined).	39
Analysis 1.5. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 5 Depressive	0,2
symptomatology at 8, 16, 24 weeks postpartum (EPDS > 12).	41
Analysis 1.6. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 6 Maternal health	
service contact at final study assessment.	42
Analysis 1.7. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 7 Maternal-infant	12
attachment at 8, 16, 24 weeks postpartum.	43
Analysis 1.8. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 8 Maternal attitudes	чJ
toward motherhood at 8, 16, 24 weeks postpartum.	44
Analysis 1.9. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 9 Maternal anxiety at 8,	
16, 24 weeks postpartum	44
Analysis 1.10. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 10 Competence in	44
	45
mothering at 8 and 16 weeks postpartum.	4)
Analysis 1.11. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 11 General physical	46
and mental health (SF-36) at final study assessment.	46
Analysis 1.12. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 12 Perceived social	(0
support at 8, 16, 24 weeks postpartum.	48
Analysis 1.13. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 13 Breastfeeding	10
duration at 8, 16, 24 weeks postpartum.	49
Analysis 1.14. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 14 Infant	
immunisations.	50
Analysis 1.15. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 15 Infant injuries.	50
Analysis 1.16. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 16 Marital discord at 4	
and 24 weeks postpartum	51
Analysis 2.1. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 1	
Antenatal and postnatal classes	51
Analysis 2.2. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 2	
Postpartum professional-based home visits	52
Analysis 2.3. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 3	
Postpartum lay-based home visits	53
Psychosocial and psychological interventions for preventing postpartum depression (Review)	i

Analysis 2.4. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 4 Early
postpartum follow up
Analysis 2.5. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 5
Continuity of care
Analysis 3.1. Comparison 3 Psychological Interventions versus usual care - variations in intervention type, Outcome 1
Psychological debriefing
Analysis 3.2. Comparison 3 Psychological Interventions versus usual care - variations in intervention type, Outcome 2
Interpersonal psychotherapy
Analysis 4.1. Comparison 4 All interventions versus usual care - variations in intervention mode, Outcome 1 Individually
based interventions
Analysis 4.2. Comparison 4 All interventions versus usual care - variations in intervention mode, Outcome 2 Group-based
interventions
Analysis 5.1. Comparison 5 All interventions versus usual care - variations in intervention onset, Outcome 1 Interventions
with antenatal and postnatal components
Analysis 5.2. Comparison 5 All interventions versus usual care - variations in intervention onset, Outcome 2 Interventions
with postnatal only component
Analysis 6.1. Comparison 6 All interventions versus usual care - variations in intervention duration, Outcome 1 Single-
contact interventions
Analysis 6.2. Comparison 6 All interventions versus usual care - variations in intervention duration, Outcome 2 Multiple-
contact interventions
Analysis 7.1. Comparison 7 All interventions versus usual care - variations in risk status, Outcome 1 Interventions for at-
risk women
Analysis 7.2. Comparison 7 All interventions versus usual care - variations in risk status, Outcome 2 Interventions for
general population
WHAT'S NEW
HISTORY
CONTRIBUTIONS OF AUTHORS
DECLARATIONS OF INTEREST
SOURCES OF SUPPORT
NOTES
INDEX TERMS

[Intervention Review]

Psychosocial and psychological interventions for preventing postpartum depression

Cindy-Lee Dennis¹, Debra K Creedy²

¹Lawrence S. Bloomberg Faculty of Nursing, University of Toronto, Toronto, Canada. ²Faculty of Nursing and Health, Griffith University, Nathan, Australia

Contact address: Cindy-Lee Dennis, Lawrence S. Bloomberg Faculty of Nursing, University of Toronto, 155 College Street, Toronto, Ontario, M5T 1P8, Canada. cindylee.dennis@utoronto.ca.

Editorial group: Cochrane Pregnancy and Childbirth Group. **Publication status and date:** Edited (no change to conclusions), published in Issue 4, 2008. **Review content assessed as up-to-date:** 15 August 2004.

Citation: Dennis CL, Creedy DK. Psychosocial and psychological interventions for preventing postpartum depression. *Cochrane Database of Systematic Reviews* 2004, Issue 4. Art. No.: CD001134. DOI: 10.1002/14651858.CD001134.pub2.

Copyright © 2008 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

ABSTRACT

Background

The cause of postpartum depression remains unclear, with extensive research suggesting a multi-factorial aetiology. However, epidemiological studies and meta-analyses of predictive studies have consistently demonstrated the importance of psychosocial and psychological variables. While interventions based on these variables may be effective treatment strategies, theoretically they may also be used in pregnancy and the early postpartum period to prevent postpartum depression.

Objectives

Primary: to assess the effect of diverse psychosocial and psychological interventions compared with usual antepartum, intrapartum, or postpartum care to reduce the risk of developing postpartum depression. Secondary: to examine (1) the effectiveness of specific types of psychosocial and psychological interventions, (2) the effectiveness of individual versus group-based interventions, (3) the effects of intervention onset and duration, and (4) whether interventions are more effective in women selected with specific risk factors.

Search methods

We searched the Cochrane Pregnancy and Childbirth Group trials register (January 27 2004), the Cochrane Depression, Anxiety and Neurosis Group trials register (October 2003), the Cochrane Central Register of Controlled Trials (October 2003), MEDLINE (1966 to 2004), EMBASE (1980 to 2004) and CINAHL (1982 to 2004). We scanned secondary references and contacted experts in the field.

Selection criteria

All published and unpublished randomised controlled trials of acceptable quality comparing a psychosocial or psychological intervention with usual antenatal, intrapartum, or postpartum care.

Data collection and analysis

Both reviewers participated in the evaluation of methodological quality and data extraction. Additional information was sought from several trial researchers. Results are presented using relative risk for categorical data and weighted mean difference for continuous data.

Main results

Fifteen trials, involving over 7600 women, were included. Overall, women who received a psychosocial intervention were equally likely to develop postpartum depression as those receiving standard care (relative risk (RR) 0.81, 95% confidence interval (CI) 0.65 to 1.02). One promising intervention appears to be the provision of intensive postpartum support provided by public health nurses or midwives (RR 0.68, 95% CI 0.55 to 0.84). Identifying mothers 'at-risk' assisted the prevention of postpartum depression (RR 0.67, 95% CI 0.51 to 0.89). Interventions with only a postnatal component appeared to be more beneficial (RR 0.76, 95% CI 0.58 to 0.98) than interventions that also incorporated an antenatal component. While individually-based interventions may be more effective (RR 0.76, 95% CI 0.59 to 1.00) than those that are group-based, women who received multiple-contact intervention were just as likely to experience postpartum depression as those who received a single-contact intervention.

Authors' conclusions

Overall psychosocial interventions do not reduce the numbers of women who develop postpartum depression. However, a promising intervention is the provision of intensive, professionally-based postpartum support.

PLAIN LANGUAGE SUMMARY

Psychosocial and psychological interventions for preventing postpartum depression

Psychosocial and psychological interventions compared with usual care provided antenatally or postnatally do not reduce the risk of postpartum depression.

Postpartum depression affects approximately 13% of all new mothers. While no clear beneficial effect in the prevention of postpartum depression from a range of psychosocial and psychological interventions was found, intensive professionally-based postpartum support may be helpful. Interventions that were individually based appear to be more beneficial than those that were group-based. There is also evidence supporting interventions that are initiated in the postnatal period that do not include an antenatal component. Finally, interventions targeting 'at-risk' mothers may be more beneficial than those including a general maternal population. Many questions remain unanswered and additional research is needed.

BACKGROUND

Postpartum mood disorders are a common form of maternal morbidity following delivery (Stocky 2000). These affective disorders range in severity from the mild and transient 'baby blues' experienced by 50% to 80% of women to postpartum psychosis, a serious condition which affects less than 1% of mothers and usually requires hospitalisation (Evins 1997). Among these disorders is postpartum depression, a condition often exhibiting the disabling symptoms of uneasiness, irritability, confusion and forgetfulness, anhedonia, fatigue, insomnia, anxiety, guilt, inability to cope, and thoughts of suicide. Frequently exacerbating these symptoms are low self-esteem, lack of confidence, and unrealistic expectations of motherhood. The development of postpartum depression is greatest in the first three months postpartum with duration frequently dependent on severity (Cox 1993). Some residual depressive symptoms are common up to a year after delivery (Cooper 1998).

Postpartum depression is a major health issue for many women from diverse cultures (Affonso 2000). Longitudinal and epidemiological studies have yielded varying prevalence rates, ranging from 3% to more than 25% of women in the first year following delivery; these rates fluctuate due to sampling, timing of assessment, differing diagnostic criteria (major or minor depression), and whether the studies were retrospective (low rates) or prospective (6- to 10fold higher). Frequently cited estimates range between 10% to 15% and a meta-analysis of 58 studies reported the prevalence of postpartum depression to be 13% (O'Hara 1996). It is noteworthy that the absolute difference in estimates between self-report assessments of depressive symptoms, such as the commonly used Edinburgh Postnatal Depression Scale (which does not diagnose postpartum depression), and standardised diagnostic interviews (which do diagnose postpartum depression) was small.

This morbidity has well documented public health consequences for the mother, child, and family. While women who have suf-

fered from postpartum depression are twice as likely to experience future episodes of depression over a five-year period (Cooper 1995), infants and children are particularly vulnerable. Postpartum depression can cause impaired maternal-infant interactions (Murray 1996) and negative perceptions of infant behaviour (Mayberry 1993), which have been linked to attachment insecurity (Hipwell 2000; Murray 1992), cognitive developmental delay (Cogill 1986; Hipwell 2000) and social/interaction difficulties (Cummings 1994; Murray 1999). Infants as young as three months of age have been shown to ably detect their mothers' mood and to modify their own responses accordingly (Cohn 1983). While cognitive skills (Whiffen 1989), expressive language development (Cox 1987), and attention (Breznitz 1988) have been negatively influenced by postpartum depression, it has also been reported that children of depressed mothers are two to five times more likely to develop long-term behavioural problems (Beck 1999; Orvaschel 1988). Child neglect/abuse (Buist 1998) and marital stress resulting in separation or divorce (Boyce 1994; Holden 1991) are other reported outcomes. Maternal and infant mortality are rare but real consequences of postpartum depression.

The aetiology of postpartum depression remains unclear and there is little evidence to support a biological basis (Beck 2001; O'Hara 1997). Despite considerable research, no single causative factor has been isolated. However, consistent findings suggest the importance of psychosocial variables (Cooper 1998; O'Hara 1997). In particular, stressful life events (Bernazzani 1997; O'Hara 1991), marital conflict (Bernazzani 1997; O'Hara 1991; O'Hara 1986), and the lack of social support (Bernazzani 1997; Brugha 1998; Cooper 1998; O'Hara 1986; Small 1994; Stein 1989; Stuchbery 1998) have been found to significantly increase the risk of postpartum depression. The saliency of social support was especially highlighted in a predictive study of several thousand women, in which mothers who lacked social support were approximately two times more likely to develop postpartum depression than mothers with sufficient support (Cooper 1996).

To address this issue, a variety of psychosocial and psychological interventions have been developed to treat postpartum depression. For example, randomised controlled trials evaluating cognitive-behavioural counselling with antidepressants (Appleby 1997), cognitive-behavioural therapy and non-directive counselling (Cooper 1997; Cooper 2003), health visitor-led non-directive counselling (Holden 1989; Wickberg 1996), peer support (Dennis 2003), and interpersonal psychotherapy (O'Hara 2000) have all demonstrated the amenability of postpartum depression to treatment.

It is theoretically plausible that psychosocial and psychological interventions may prevent postpartum depression, as many of the known risk factors are present during pregnancy and the immediate postpartum period. Furthermore, a number of studies have suggested that there is an overlap between antenatal and postpartum depression, in that there are significant correlations among Edinburgh Postnatal Depression Scale scores at varying antenatal and immediate postnatal time periods (Appleby 1994; Dennis 2004b; Hannah 1992; Lane 1997; Yamashita 2000). While psychosocial and psychological interventions may be effective treatment strategies, they may also be used in pregnancy and the early postpartum period to prevent postpartum depression.

OBJECTIVES

The primary objective of this review was to assess the effects, on mothers and their families, of preventive psychosocial and psychological interventions compared with usual antepartum, intrapartum, or postpartum care to reduce the risk of postpartum depression.

Secondary objectives were to examine:

1. the effectiveness of specific types of psychosocial interventions;

2. the effectiveness of specific types of psychological interventions;

3. the effects of intervention mode (e.g. individual versus group-based interventions);

4. the effects of intervention onset (e.g. antenatal and postnatal interventions versus postnatal only interventions);

5. the effects of intervention duration (e.g. single-contact interventions versus multiple-contact interventions);

6. the effects of sample selection criteria (e.g. targeting women with specific risk factors versus the general population).

METHODS

Criteria for considering studies for this review

Types of studies

All published, unpublished and ongoing randomised controlled trials of preventive psychosocial or psychological interventions in which the primary or secondary aim was reduction in risk to develop postpartum depression. Quasi-randomised trials (e.g., those randomised by delivery date, or odd versus even medical record numbers) were excluded from the analysis.

Types of participants

Pregnant women and new (less than six weeks postpartum) mothers, including those at no known risk and those identified as atrisk to develop postpartum depression.

Types of interventions

Any form of standard or usual care compared to a variety of non-pharmaceutical interventions - including psychoeducational strategies, cognitive behavioural therapy, interpersonal psychotherapy, non-directive counselling, psychological debriefing, various supportive interactions, and tangible assistance - delivered via telephone, home or clinic visits, or individual or group sessions antenatally and/or within the first month postpartum by a professional (nurse, midwife, childbirth educator, physician) or lay person (a specially trained woman from the community, a student).

Types of outcome measures

A. Maternal outcomes

1. Postpartum depression (as variously defined and measured by trialists)

2. Postpartum psychosis

3. Maternal mortality and serious morbidity including self-harm, suicide attempts

4. Health service utilisation including outpatient and inpatient use of psychiatric unit, other health services

- 5. Maternal-infant attachment
- 6. Maternal attitudes towards motherhood
- 7. Anxiety
- 8. Stress
- 9. Maternal confidence
- 10. Maternal competence
- 11. Self-esteem
- 12. General health
- 13. Maternal dissatisfaction with intervention
- 14. Maternal perceived social support

B. Infant outcomes

15. Breastfeeding duration (variously defined)

16. Breastfeeding level (exclusive, almost exclusive, high, partial, token, bottle-feeding)

- 17. Infant health parameters including immunisation, accidental injury, non accidental injury
- 18. Infant developmental assessments (variously defined)
- 19. Child abuse and/or neglect
- 20. Neonatal/infant mortality
- 21. Neonatal/infant morbidity
- 22. Quality of mothering (variously defined)

C. Family outcomes

- 23. Marital discord
- 24. Marital separation/divorce

Search methods for identification of studies

Electronic searches

We searched the Cochrane Pregnancy and Childbirth Group trials register by contacting the Trials Search Co-ordinator (January 27 2004).

The Cochrane Pregnancy and Childbirth Group's trials register is maintained by the Trials Search Co-ordinator and contains trials identified from:

1. quarterly searches of the Cochrane Central Register of Controlled Trials (CENTRAL);

2. monthly searches of MEDLINE;

3. hand searches of 30 journals and the proceedings of major conferences;

4. weekly current awareness search of a further 37 journals. Details of the search strategies for CENTRAL and MEDLINE, the list of hand-searched journals and conference proceedings, and the list of journals reviewed via the current awareness service can be found in the 'Search strategies for identification of studies' section within the editorial information about the Cochrane Pregnancy and Childbirth Group.

Trials identified through the searching activities described above are given a code (or codes) depending on the topic. The codes are linked to review topics. The Trials Search Co-ordinator searches the register for each review using these codes rather than keywords. In addition, we searched the Cochrane Depression, Anxiety and Neurosis trials register (October 2003), the Cochrane Central Register of Controlled Trials (October 2003), MEDLINE (1966 to 2004), EMBASE (1980 to 2004) and CINAHL (1982 to 2004) using various combinations of the terms postpartum/postnatal depression. We scanned secondary references and obtained promising studies and made contacts with experts in the field to identify other published or unpublished trials.

Data collection and analysis

Selection of trials

Titles and abstracts of the electronic searches were reviewed by the primary reviewer. We independently evaluated trials under consideration for methodological quality and appropriateness for inclusion, without consideration of their results. We resolved uncertainties regarding the appropriateness for inclusion through discussion and consensus.

Methodological quality assessment

We assessed the quality of the trials that met the eligibility criteria using the following criteria:

1. generation of random allocation sequence: adequate, inadequate, unclear;

2. allocation concealment: A = adequate, B = unclear, C = inadequate;

3. blinding of participants: yes, no, inadequate, no information;

4. blinding of caregivers: yes, no, inadequate, no information;

5. blinding of outcome assessment: yes, no, inadequate or no information;

6. completeness of follow-up data (including any differential loss of participants from each group): A = less than 3% of participants excluded, B = 3% to 9.9% of participants excluded, C = 10% to 19.9% excluded, D = 20% or more excluded, E = unclear;

7. analysis of participants in randomised groups.

We assigned a rating to each trial, compared results and discussed differences until we reached agreement. We have clearly described reasons for exclusion of any apparently eligible trial (*see* 'Characteristics of excluded studies' table).

Data extraction

We independently extracted data from trial reports using a pilottested data extraction form developed by the primary reviewer. Wherever necessary, we requested unpublished or missing data from the trial contact author. In addition, we sought data to allow an 'intention-to-treat' analysis. Data were entered into RevMan 2000 by one reviewer and double data entry was completed by the other reviewer or a research assistant.

Data synthesis

Trials using different preventive strategies were analysed separately and the results combined only if there was no reason to think that they differed in relevant ways. While the primary meta-analysis was based on the occurrence of postpartum depression or not (however measured by trialists), we incorporated several depression rating scales or cut-off points. To address the potential measurement differences, we used a fixed effect model to make direct comparisons between trials using the same rating scale and cutoff. If trials used different ways of measuring the same continuous outcome, we used standardised mean differences. We performed meta-analyses using relative risks as the measure of effect size for binary outcomes, and weighted mean differences for continuous outcome measures, both with 95% confidence intervals. We assessed the extent to which there were between-study differences including variations in the population or intervention.

We used fixed effect meta-analysis to combine study data. We investigated heterogeneity by calculating I² statistics (Higgins 2002), and if this indicated a high level of heterogeneity among the trials included in an analysis (I² > 50%), we used random effects metaanalysis for an overall summary. Where we found high levels of heterogeneity, we explored these by sensitivity analyses excluding the trials most susceptible to bias based on the following quality assessment: (1) those with unclear allocation concealment (B); (2) high levels of postrandomisation losses or exclusions (D); or (3) unblinded outcome assessment or blinding of outcome assessment uncertain.

Subgroup analyses

We planned and completed the following six a priori subgroup analyses:

1. the effectiveness of specific types of psychosocial interventions:

2. the effectiveness of specific types of psychological interventions;

3. the effects of intervention mode (e.g. individual versus group-based interventions);

4. the effects of intervention onset (e.g. antenatal and postnatal interventions versus postnatal only interventions);

5. the effects of intervention duration (e.g. single-contact interventions versus multiple-contact interventions);

6. the effects of sample selection criteria (e.g. women with specific risk factors versus the general population).

RESULTS

Description of studies

See: Characteristics of included studies; Characteristics of excluded studies; Characteristics of ongoing studies.

Please see table of 'Characteristics of included studies'. Fifteen trials, reported between 1995 and 2003 and including 7697 women, were identified and met the inclusion criteria. The trials were primarily conducted in Australia and the UK; two trials were conducted in the USA (Gorman 2002; Zlotnick 2001) and one was conducted in China (Tam 2003). While all trials included the outcome postpartum depression, several studies provided data on other variables including: health service contact (Brugha 2000), maternal-infant attachment (Armstrong 1999), maternal attitudes towards motherhood (Armstrong 1999), anxiety (Lavender 1998), competence in mothering (Armstrong 1999), general physical and mental health (Gunn 1998; Morrell 2000; Reid 2002; Small 2000), perceived support (Morrell 2000; Reid 2002), breastfeeding duration (Armstrong 1999; Gunn 1998; Morrell 2000), infant immunisation (Armstrong 1999), infant injury (Armstrong 1999), and marital discord (Gorman 2002).

Definition of postpartum depression

In all trials but one (Zlotnick 2001), postpartum depressive symptomatology was defined as a score above a specified cut-off point on a self-report measure; for the majority of studies (10 out of 15) an Edinburgh Postnatal Depression Scale (EPDS) score greater than 12 (also reported as a 12/13 cut-off score) indicated postpartum depression. Several studies also reported mean EPDS scores (Armstrong 1999; Gorman 2002; Gunn 1998; MacArthur 2002; Morrell 2000; Reid 2002; Small 2000). Two additional trials used the EPDS to measure postpartum depression but incorporated a different cut-off score; Brugha 2000 used a 10/11 cut-off while Reid 2002 selected a 11/12 cut-off. It is important to note that the EPDS does not diagnose postpartum depression (as this can only be accomplished through a psychiatric clinical interview) but rather it is the most frequently used instrument to assess for postpartum depressive symptomatology. Created to counter the limitations of other well-established depression scales, the EPDS has been validated by standardised psychiatric interviews with large samples and has well-documented reliability and validity in over 11 languages. Two trials used a self-report measure other than the EPDS (Lavender 1998; Tam 2003); both used the Hospital Anxiety Depression Scale. Both Gorman 2002 and Zlotnick 2001 used a semi-structured diagnostic interview (Structured Clinical Interview for DSM-IV) to assess for depression.

The timing of the outcome assessment varied considerably between studies, ranging from 3 (Lavender 1998) to 24 (Gorman 2002; Gunn 1998; Morrell 2000; Priest 2003; Reid 2002; Small 2000; Stamp 1995) weeks postpartum; one trial also included a 52-week assessment (Priest 2003).

Types of psychosocial interventions

The studies were subgrouped into categories to examine specific types of psychosocial interventions such as antenatal and postnatal classes (Brugha 2000; Reid 2002; Stamp 1995), professional (Armstrong 1999; MacArthur 2002) and lay (Morrell 2000) home visits, continuity of care (Waldenstrom 2000), and early postpartum follow up (e.g. routine postpartum care initiated earlier than standard practice) (Gunn 1998). The interventions were provided by a variety of professionals including nurses (Armstrong 1999; Brugha 2000), physicians (Gunn 1998), midwives (MacArthur 2002; Reid 2002; Stamp 1995; Waldenstrom 2000), and allied healthcare providers (e.g. occupational therapist) (Brugha 2000). In one trial the intervention was provided by lay individuals (Morrell 2000). In the majority of studies, the control group was reported to have received usual antenatal/postnatal care, which varied both between and within countries. Wherever there were individual study details on care received by the control group, these are presented in the table of included studies.

Types of psychological interventions

The studies were subgrouped into categories to examine specific types of psychological interventions, such as debriefing (Gamble 2003; Lavender 1998; Priest 2003; Small 2000; Tam 2003) and interpersonal psychotherapy (Gorman 2002; Zlotnick 2001). The interventions were provided by diverse healthcare professionals including midwives (Gamble 2003; Lavender 1998; Priest 2003; Small 2000), nurses (Tam 2003), and mental health specialists (Gorman 2002).

Other health outcomes

Reporting of other maternal health outcomes was inconsistent across studies; the main exception was the use of the SF-36 by four trials to examine general physical and mental health (Gunn 1998; Morrell 2000; Reid 2002; Small 2000). One study reported infant health outcomes (Armstrong 1999) and another included the family outcome of 'marital discord' (Gorman 2002).

Differences in groups studied

Seven trials targeted high-risk women based on various factors believed to put them at additional risk of postpartum depression (Armstrong 1999; Brugha 2000; Gamble 2003; Gorman 2002; Stamp 1995; Tam 2003; Zlotnick 2001), while the other eight trials enrolled women from the general population.

Risk of bias in included studies

Randomisation was performed most frequently by consecutively numbered, sealed, opaque envelopes (Gamble 2003; Lavender 1998; Morrell 2000; Priest 2003; Stamp 1995; Tam 2003; Waldenstrom 2000). Various forms of computer-based randomisation was used by four trials (Armstrong 1999; Brugha 2000; MacArthur 2002; Reid 2002). Two trials incorporated a central, computerised randomisation service accessed by telephone (Gunn 1998; Small 2000) and one trial used a block randomisation procedure using a random numbers table (Gorman 2002). Allocation concealment was unclear in one trial (Zlotnick 2001). A power analysis was completed by all but two trials (Gorman 2002; Zlotnick 2001) and data were analysed using an intent-to-treat approach. Outcome data were collected by assessors blinded to group allocation (Armstrong 1999; Brugha 2000; Gorman 2002) or mailed questionnaires; for one study the identity of the outcome assessor (Zlotnick 2001). Five trials had a follow-up attrition rate greater than 20%: Gunn 1998 (34% at 24 weeks); MacArthur 2002 (27% at 16 weeks); Morrell 2000 (21% at 24 weeks); Reid 2002 (29% at 24 weeks); and Tam 2003 (21% at 6 weeks). It is noteworthy that follow up in all these trials included mailed questionnaires. Based on susceptibility to bias (e.g. unclear allocation concealment, high levels of postrandomisation losses or exclusions, or unblinded outcome assessment), the following trials

were excluded as appropriate during the sensitivity analysis for outcomes with high levels of heterogeneity ($I^2 > 50\%$): Gunn 1998; MacArthur 2002; Morrell 2000; Reid 2002; Tam 2003; Zlotnick 2001.

Effects of interventions

Fifteen trials, involving over 7600 women, were included. The results are presented in sequential order, starting with maternal outcomes followed by infant and family outcomes. Because of the large number of maternal outcomes in this Review, the following summary of results is restricted to data collected and reported in at least two trials. Please refer to the meta-analyses graphs for the full results. The meta-analyses for several outcomes had significant heterogeneity. However, the removal of trials at risk of bias resulted in no substantial changes to any of the conclusions. All sensitivity analyses are presented in the meta-analyses graphs. Outcomes that were assessed at 8, 16, and 24 weeks were categorised and presented in the results as follows:

- 1. 0 to 8 weeks short-term effects;
- 2. 9 to 16 weeks intermediate effects;
- 3. 17 to 24 weeks longer-term effects.

Comparison one (main comparison): All psychosocial and psychological interventions versus usual care - all trials

A. Maternal outcomes

We considered 14 maternal outcomes. Data were not available for the following prespecified outcomes: postpartum psychosis, mortality, maternal stress, maternal confidence, self-esteem, and dissatisfaction with intervention.

Outcome: Depressive symptomatology at last assessment (variously defined)

The main outcome measure for this Review was postpartum depression at last study assessment. There was no beneficial effect on the prevention of postpartum depression in the meta-analysis of all types of interventions (15 trials, n = 7697; relative risk (RR) = 0.81, 95% confidence interval (CI) 0.65 to 1.02). There was significant heterogeneity among these trials (I² = 68.8%). A similar non-significant effect was found when weighted mean differences (WMD) were calculated among the trials that provided mean scores (8 trials, n = 4880; WMD = -0.36, 95% CI -1.21 to 0.48).

Outcome: Depressive symptomatology at last assessment (Edinburgh Postnatal Depression Scale (EPDS) greater than 12)

To address potential measurement differences, a direct comparison using a random effects model was made between trials that used the same rating scale. For this Review, the most commonly used measure to assess depressive symptoms was the Edinburgh Postnatal Depression Scale, employing the recommended 12/13 cut-off score. Similar to the meta-analysis incorporating all measures, no preventive effect was found when all psychosocial and psychological interventions were grouped together (10 trials, n = 6126; RR = 0.91, 95% CI 0.73 to 1.15).

Outcome: Depressive symptomatology at 8, 16, and 24 weeks (variously defined)

Results suggested a short-term reduction in depressive symptomatology (8 trials; n = 4091; RR = 0.65, 95% CI 0.43 to 1.00). However the effects appeared to weaken at the intermediate period (8 trials, n = 3326; RR = 0.80, 95% CI 0.56 to 1.12) and disappear when measured later (7 trials, n = 4314; RR = 1.02, 95% CI 0.87 to 1.19) in the postpartum period.

Outcome: Depressive symptomatology at 8, 16, and 24 weeks (defined as EPDS > 12)

When only trials that used the EPDS > 12 as the outcome measure were included, no apparent short-term benefits were found (6 trials, n = 3452; RR = 0.90, 95% CI 0.65 to 1.25). Similar results were found when depressive symptomatology was assessed at the intermediate (5 trials, n = 2369; RR = 0.72, 95% CI 0.49 to 1.06) and longer-term (6 trials, n = 3598; RR = 1.00, 95% CI 0.84 to 1.19) time periods.

Outcome: Maternal physical and mental health (SF-36) at last study assessment

We found no apparent effects among any of the scale subcategories: physical functioning (4 trials; n = 2589; WMD = -.29, 95% CI -0.91 to 1.49); physical role functioning (4 trials; n = 2588; WMD = -.90, 95% CI -3.33 to 1.52); bodily pain (4 trials; n = 2589; WMD = .25, 95% CI -1.41 to 1.92); mental health (4 trials; n = 2582; WMD = -.85, 95% CI -2.21 to 0.52); emotional role functioning (4 trials; n = 2586; WMD = -.93, 95% CI -3.55 to 1.69); vitality (4 trials; n = 2581; WMD = .64, 95% CI -.99 to 2.28); social functioning (4 trials; n = 2591; WMD = -.59, 95% CI -2.29 to 1.10); and general health (4 trials; n = 2586; WMD = -.19, 95% CI -1.68 to 1.29).

Outcome: Perceived social support at 8, 16, and 24 weeks

Two trials measured maternal perceptions of support at 24 weeks using different measures; no beneficial effect was demonstrated (2

trials, n = 1174; standardised mean difference = 0.02, 95% CI - 0.09 to 0.14).

B. Infant outcomes

Outcome: Breastfeeding duration

Three trials examined breastfeeding duration and found no short-term (n = 722; RR = 1.03, 95% CI 0.89 to 1.19) or longer-term (n = 968; RR = 0.90, 95% CI 0.74 to 1.10) effects.

Other outcomes

Only one trial reported on other infant outcomes. The mean number of immunisations infants received at three to four months was higher (n = 160; WMD = 0.42, 95% CI 0.11 to 0.73) and the likelihood of infant injuries was lower (n = 160; RR = 0.54, 95% CI 0.31 to 0.92) in the intervention group.

C. Family outcomes

Only one trial reported on family outcomes. There was no significant effect on marital discord scores at four weeks (n = 31; WMD = -3.20, 95% CI -16.93 to 10.53) and 24 weeks (n = 29; WMD = -7.90, 95% CI -21.52 to 5.72) postpartum.

Comparison two: Impact of various types of psychosocial interventions

We found no preventive effect when the interventions were antenatal and postnatal classes (2 trials, n = 311; RR = 1.02, 95% CI 0.61 to 1.72), lay home visits (1 trial, n = 481; RR = 0.89, 95% CI 0.62 to 1.27), early postpartum follow-up (1 trial, n = 475; RR = 0.91, 95% CI 0.56 to 1.48), or continuity of care (1 trial, n = 935; RR = 1.34, 95% CI 0.97 to 1.85). However, we found a beneficial effect when the intervention involved home visits by a health professional (2 trials, n = 1663; RR = 0.68, 95% CI 0.55 to 0.84).

Comparison three: Impact of various types of psychological interventions

We found no preventive effect when the intervention was psychological debriefing (5 trials, n = 3051; RR = 0.57, 95% CI 0.31 to 1.04) or interpersonal psychotherapy (2 trials, n = 72; RR = 0.31, 95% CI 0.04 to 2.52).

Comparison four: Influence of variations in mode of delivery

Outcome: Individually-based interventions

Analysis of 11 trials of interventions provided to individual women suggested a possible benefit in preventing the number of women with depressive symptomatology at the last study assessment (n = 6642; RR = 0.76, 95% CI 0.59 to 1.00). When trials susceptible to bias were removed, the direction of the effect remained the same but the 95% confidence interval widened (7 trials, n = 3667; RR = 0.68, 95% CI 0.43 to 1.09). A similar trend was found when depressive outcomes were assessed within 0 to 8 weeks postpartum (7 trials, n = 3963; RR = 0.64, 95% CI 0.40 to 1.01). However, no clear beneficial effect was found at 9 to 16 weeks postpartum (4 trials, n = 2241; RR = 0.71, 95% CI 0.45 to 1.12), and 17 to 24 weeks postpartum (5 trials, n = 3484; RR = 0.98, 95% CI 0.82 to 1.17).

Outcome: Group-based interventions

Of the four trials evaluating interventions delivered to groups of women, there was no apparent reduction in depressive symptomatology at last study assessment (n = 1055; RR = 1.03, 95% CI 0.65 to 1.63). Analyses according to timing of measurement indicate no apparent short-term (1 trial, n = 128; RR = 0.73, 95% CI 0.31 to 1.69), intermediate (4 trials, n = 1085; RR = 0.93, 95% CI 0.54 to 1.59), or longer-term (2 trials, n = 830; RR = 1.20, 95% CI 0.85 to 1.71) effects.

Comparison five: Influence of intervention onset

Studies in which the intervention began antenatally and continued postnatally failed to reduce the likelihood of postpartum depressive symptomatology (4 trials, n = 1283; RR = 1.21, 95% CI 0.93 to 1.59). However, a preventive effect was found for those trials evaluating a postnatal-only intervention (10 trials, n = 6379; RR = 0.76, 95% CI 0.58 to 0.98).

Comparison six: Influence of intervention duration

In the four trials that evaluated a single-contact intervention (e.g. psychological debriefing, early postpartum follow up) the relative risk was 0.70 (n = 2898; 95% CI 0.38 to 1.27). In the 11 trials in which the intervention involved multiple contacts the relative risk was 0.84 (n = 4790; 95% CI 0.66 to 1.08).

Comparison seven: Influence of sample selected

Trials selecting participants based on 'at-risk' criteria had more apparent success in preventing postpartum depression (7 trials, n = 1162; RR = 0.67, 95% CI 0.51 to 0.89) than the trials that enrolled women from the general population (8 trials, n = 6535; RR = 0.87, 95% CI 0.66 to 1.16).

DISCUSSION

This Review summarises the results of 15 trials involving 7697 women, that were conducted in four countries under a wide variety of circumstances. The methodological quality of the included trials was good. All but one trial (Morrell 2000) involved a psychosocial or psychological intervention provided by a health professional. However, the reporting of the trials was often not comprehensive, lacking in terms of details in the training and qualifications of the intervention providers and in the description of adherence to the intervention protocol. There was also a failure to present details of the informational element of the interventions and on the background features of the care received by the control groups. While intent-to-treat data analyses were performed, trials involving group sessions had high levels of non-compliance with group attendance (Brugha 2000; Reid 2002; Stamp 1995). The removal of trials at risk of bias resulted in no substantial changes to any of the conclusions.

In the primary comparison, the diversity of preventive interventions and the widely differing study end-points should urge some caution in the interpretation of the pooled data. To partially address this issue, the meta-analyses included short-, intermediate-, and longer-term effects where appropriate. Despite this caution and the subgrouping of end-points, this Review consistently demonstrated that women who received a preventive intervention were overall just as likely to experience postpartum depression as those who received standard care. It is unknown to what extent some of the heterogeneity or insignificant results seen in this Review are related to the measure used to assess postpartum depression. However, a similar non-significant effect was found among those trials that incorporated the widely used Edinburgh Postnatal Depression Scale to measure depressive symptomatology.

In general, the effectiveness of psychosocial or psychological approaches has not been demonstrated. Antenatal classes focusing on postpartum depression have repeatedly been shown to have no preventive effect and cannot be recommended at this time. Similarly, the trials evaluating in-hospital psychological debriefing provide good evidence to suggest that this intervention should not be implemented into practice. The effectiveness of interpersonal psychotherapy and lay support remains uncertain. Morrell 2000 demonstrated that the addition of home visits by a community support worker had no protective effect on postpartum depression. However, a review of the intervention activities revealed that the lay women spent a significant amount of their time providing instrumental support, such as housework and infant care, and limited time providing emotional and appraisal (feedback) support to the mother. The potential to positively influence health outcomes depends on predicting which supportive functions will be the most effective for a particular type of stressor (Will 2000). In qualitative studies, women from diverse cultures who have suffered from postpartum depression consistently describe their feelings of loneliness, worries about maternal competence, role conflicts, and inability to cope (Chen 1999; Nahas 1999; Ritter 2000; Small 1994); apparently the presence or absence of instrumental support was not a factor.

Improving the quality of care provided to women has been another postpartum depression preventive approach. Two trials have evaluated the effect of early postpartum follow up. Although one quasi-experimental study was not included in this Review (Serwint 1991), another well-designed trial demonstrated no beneficial effect on maternal mental health outcomes (Gunn 1998). As such, there is preliminary evidence to suggest that early postpartum follow up has no preventive effect on postpartum depression and cannot be recommended for clinical practice. Similar results have been found with midwifery-based continuity of care models (Waldenstrom 2000).

However, there is beginning evidence to suggest the importance of additional professional support provided postnatally. While one well-designed trial (Armstrong 1999) suggested intensive nursing home visits with at-risk mothers was protective during the first six weeks postpartum, the beneficial effect was not maintained to 16 weeks. It is noteworthy that the 16-week assessment coincided with a decrease in intervention intensity from weekly to monthly nursing visits. Results from a cluster randomised controlled trial demonstrated that flexible, individualised midwifery-based postpartum care that incorporated postpartum depression screening tools also had a preventive effect (MacArthur 2002).

While there was diversity in the types of intervention provided, the trials included in this Review incorporated a primary preventive intervention; no trial selected participants based on evidence of depressive symptomatology. According to Shah 1998, preventive interventions incorporate any strategy that (1) reduces the likelihood of a disease/condition affecting an individual (primary prevention); (2) interrupts or slows the progress of a disease/condition through early detection and treatment (secondary prevention); or (3) slows the progress of a disease/condition and reduces resultant disability through treatment of established disease (tertiary prevention). These preventive interventions can be further classified into different categories depending on the target population: (1) universal interventions are designed to be offered to all women; (2) selective interventions are designed to be offered to women at increased risk of developing depression; and (3) indicated interventions are designed to be offered to women who have been identified as depressed or probably depressed (Mrazek 1994). To examine the effects of universal and selective interventions, subgroup analyses were conducted. The results suggest identifying mothers 'at-risk' assisted in the prevention of postpartum depression. However, currently there is no consistency in the identification of women 'at-risk' and a review of 16 antenatal screening tools suggests that there are no measures with acceptable predictive validity to accurately identify women who will later develop postpartum depression (Austin 2003). This may partially explain why interventions with only a postnatal component appear to be more

beneficial than interventions that also incorporate an antenatal component. Other differences in intervention delivery were also examined. While individually based interventions may be more beneficial than those that are group-based, women who receive a multiple-contact intervention were just as likely to develop postpartum depression as those who received a single-contact intervention.

The preventive interventions had no apparent effect on other maternal outcomes, including health service contact, maternal-infant attachment, maternal attitudes towards motherhood, maternal competence, general physical and mental health, perceived support, breastfeeding duration, and marital discord. However, one study (Armstrong 1999) reported improved mean number of completed infant immunisations and decreased rates of infant injury among mothers who received intensive nursing home visits.

The long-term consequences of postpartum depression suggest preventive approaches are warranted. Manipulation of a risk factor may improve the associated likelihood of developing postpartum depression through many different ways. The most obvious is to decrease the amount of exposure to a given risk factor or, alternatively, reduce the strength or mechanism of the relationship between the risk factor and postpartum depression (McLennan 2002). However, translating risk factor research into predictive screening protocols (Austin 2003) and preventive interventions has met with limited success, as complex interactions of biopsychosocial risk factors with individual variations need to be considered. Although theoretical justifications for many of these preventive approaches were presented by the individual researchers, limited evidence is available to strongly guide practice or policy recommendations Details of research currently in progress are provided in the 'Characteristics of ongoing studies' table.

AUTHORS' CONCLUSIONS

Implications for practice

Currently, there is no evidence to recommend the following interventions be implemented into practice: antenatal and postnatal classes, lay home visits, early postpartum follow up, continuity of care models, in-hospital psychological debriefing, and interpersonal psychotherapy. However, professionally based home visits, such as intensive nursing home visits and flexible postpartum care provided by midwives, appears to show promise in the prevention of postpartum depression. It is noteworthy that the latter intervention incorporated screening with a checklist and the Edinburgh Postnatal Depression Scale (EPDS) to individualise the provision of care. Interventions that are individually based may be more beneficial than those that are group-based. There also appears to be evidence supporting interventions that are initiated in the postnatal period that do not include an antenatal component. Finally, interventions targeting 'at-risk' mothers may be more beneficial than those including a general maternal population.

Implications for research

Despite the recent upsurge of interest in this area, many questions remain unanswered.

Specific research implications

• Currently, there is no evidence to support the use of antenatal group interventions in heterogeneous samples of women 'at-risk' to develop postpartum depression. This finding may be due to methodological limitations such as inadequate sample sizes, unrealistic effect sizes or no formal justification for sample size, large rates of participant decline and/or intervention attrition rates, or lack of adequate antenatal screening tools for identification of those 'at-risk' leading to the targeting of heterogeneous samples. If additional research is conducted, structured interventions with homogeneous, symptomatic women should be evaluated; this would incorporate using an 'indicated' approach. These studies must address previous methodological limitations, such as low participation rates, and should examine the efficacy for both antenatal symptoms as well as the prevention of postpartum depression.

• Further research is warranted to examine the effectiveness of nursing home visits with a specific focus on visit content and intervention intensity.

• Flexible, individualised postnatal care provided by a professional that incorporates postpartum depression screening tools appears to be promising. A well-designed trial conducted outside a UK-midwifery context is needed to replicate the results.

• Trials evaluating individually based lay interventions specifically targeting maternal mood are required. Characteristics of the lay individuals (peers versus general community-based workers) and the nature of the relationships developed should be explored.

• The importance of psychosocial interventions in preventing minor depression (for example, EPDS score greater than nine but less than 13) has not been explored. This is particularly important since research suggests that minor depressive symptomatology often precedes a major depressive episode.

General research implications

To be most efficient in conducting this research there continues to be a need for further interdisciplinary networking among investigators with complementary research interests. For example, psychosocial intervention researchers could collaborate with health services researchers to develop and test multi-level intervention approaches embedded in service systems. To further address postpartum depression as a public health problem, the inclusion of

ethnically and socio-economically diverse women in these research efforts is critical to examining the differences in depression symptoms, response rate to interventions, and health service use. In addition, all trials should include an economic analysis of the relative costs and benefits.

It is also necessary to present a few general comments regarding the development of preventive programs. Similar to screening initiatives, preventive interventions should be relatively simple and inexpensive. This is critical if the intervention is to be applied to a relatively large population; unless a project is feasible on a large scale, there is little utility in pursuing smaller demonstration projects. Furthermore, the risk of negative outcomes from a prevention intervention is a frequently ignored possibility. Although adverse effects are primarily thought of in treatment contexts, particularly pharmacological trials, preventive interventions also include the possibility of unfavourable events. For example, targeted prevention trials carry the risk of labelling and stigmatising participants. Although these risks might be tolerable for those who are accurately identified and who benefit from the intervention, it may not be for those who were included in the intervention as false positives or who do not benefit from the intervention (McLennan 2002). In addition, an increased rate of anxiety for mothers may be of real consequence, as a link between postpartum depression and child health outcomes has been demonstrated. While emphasising this may increase a mother's willingness to accept a preventive intervention, it might also augment her level of anxiety or guilt if she perceives personal responsibility for placing her child at risk for a poor outcome, particularly if she is suffering from the cognitive distortions of depression that foster excessive guilt feelings (McLennan 2002).

ACKNOWLEDGEMENTS

As part of the pre-publication editorial process, this review has been commented on by three peers (an editor and two referees who are external to the editorial team), one or more members of the Pregnancy and Childbirth Group's international panel of consumers and the Group's Statistical Adviser.

The review authors gratefully acknowledges Josephine Kavanagh, for reviewing the draft protocol and assisting with the search strategy, and Georgie Stamp, for preparing the initial draft protocol of the review which was then assumed by Cindy-Lee Dennis. The review authors also wish to thank Dr Ellen Hodnett for her excellent editorial assistance and the following study authors who were very helpful in responding to queries: Dr L Gorman, Dr R Hagan, Dr M Hughes, and Dr C MacArthur.

REFERENCES

References to studies included in this review

Armstrong 1999 {published data only}

Armstrong K, Fraser J, Dadds M, Morris J. A randomized controlled trial of nurse home visiting to vulnerable families with newborns. *Journal of Paediatrics & Child Health* 1999; **35**(3):237–44.

Armstrong K, Fraser J, Dadds M, Morris J. Promoting secure attachment, maternal mood, and child health in a vulnerable population: a randomized controlled trial. *Journal of Paediatric Child Health* 2000;**36**:555–62.

Brugha 2000 {published data only}

Brugha T, Wheatley S, Taub N, Culverwell A, Friedman T, Kirwan P, et al.Pragmatic randomized trial of antenatal intervention to prevent post-natal depression by reducing psychosocial risk factors. *Psychological Medicine* 2000;**30**(6): 1273–81.

Gamble 2003 {published data only}

Gamble J, Creedy D. Reducing postpartum emotional distress: a randomised controlled trial. [abstract]. Perinatal Society of Australia and New Zealand. 7th Annual Congress; 2003 March 9-12; Tasmania, Australia. 2003: A29.

Gorman 2002 {published data only}

Gorman L. Prevention of postpartum depression in a high risk sample. University of Iowa, Dissertation 2001.

Gunn 1998 {published data only}

Gunn J, Lumley J, Chondros P, Young D. Does an early postnatal check-up improve maternal health: Results from a randomised trial in Australian general practice. *British Journal of Obstetrics & Gynaecology* 1998;**105**(9):991–7.

Lavender 1998 {published data only}

Lavender T, Walkinshaw S. Can midwives reduce postpartum psychological morbidity? A randomized trial. *Birth* 1998;**25**(4):215–9.

MacArthur 2002 {published and unpublished data}

MacArthur C, Winter H, Bick D, Knowles H, Liford R, Henderson C, et al.Effects of redesigned community postnatal care on women's health 4 months after birth: a cluster randomised controlled trial. *Lancet* 2002;**359** (9304):378–85.

Morrell 2000 {published data only}

Morrell C, Spiby H, Stewart P, Walters S, Morgan A. Costs and effectiveness of community postnatal support workers: randomised controlled trial. *BMJ* 2000;**321**(7261):593–8. Morrell CJ, Spiby H, Stewart P, Walters S, Morgan A. Costs and benefits of community postnatal support workers: a randomised controlled trial. *Health Technology Assessment* (*South Hampton, NY*) 2000;**4**(6):1–100.

Priest 2003 {published and unpublished data}

Priest S, Henderson J, Evans S, Hagan R. Stress debriefing

after childbirth: a randomized controlled trial. *Medical Journal of Australia* 2003;**178**:542–5.

Reid 2002 {published data only}

Reid M, Glazener C, Murray G, Taylor G. A two-centred pragmatic randomized controlled trial of two interventions for postnatal support. *BJOG: an international journal of obstetrics and gynaecology* 2002;**109**:1164–70.

Small 2000 {published data only}

Small R, Lumley J, Donohue L, Potter A, Waldenstrom U. Randomised controlled trial of midwife led debriefing to reduce maternal depression after operative childbirth. *BMJ* 2000;**321**(7268):1043–7.

Stamp 1995 {published data only}

Stamp G, Williams A, Crowther C. Evaluation of antenatal and postnatal support to overcome postnatal depression: a randomized controlled trial. *Birth* 1995;**22**(3):138–43.

Tam 2003 {published data only}

Tam WH, Lee DTS, Chiu HFK, Ma KC, Lee A, Chung TKH. A randomised controlled trial of educational counselling on management of women who have suffered suboptimal outcomes in pregnancy. *BJOG: an international journal of obstetrics and gynaecology* 2003;**110**:853–9.

Waldenstrom 2000 {published data only}

Waldenstrom U, Brown S, McLachlan H, Forster D, Brennecke S. Does team midwife care increase satisfaction with antenatal, intrapartum, and postpartum care? A randomized controlled trial. *Birth* 2000;**27**(3):156–67.

Zlotnick 2001 {published data only}

Zlotnick C, Johnson S, Miller I, Pearlstein T, Howard M. Postpartum depression in women receiving public assistance: pilot study of an interpersonal-therapy-oriented group intervention. *American Journal of Psychiatry* 2001; **158**(4):638–40.

References to studies excluded from this review

Buist 1999 {published data only}

Buist A, Westley D, Hill C. Antenatal prevention of postnatal depression. *Archives of Women's Mental Health* 1999;**1**:167–73.

Chabrol 2002 {published data only}

Chabrol H, Teissedre F, Saint-Jean M, Teisseyre N, Roge B, Mullet E. Prevention and treatment of post-partum depression: a controlled randomized study on women at risk. *Psychological Medicine* 2002;**32**(6):1039–47. Chabrol H, Teissedre F, Saint-Jean M, Teisseyre N, Sistac C, Michaud C, et al. Detection, prevention and treatment of postpartum depression: a controlled study of 859 patients. *Encephale* 2002;**28**(1):65–70.

Cooper 2002 {published data only}

Cooper PJ, Landman M, Tomlinson M, Molteno C, Swartz L, Murray L. Impact of a mother-infant intervention in an indigent peri-urban South African context: pilot study. *British Journal of Psychiatry* 2002;**180**:76–81.

D'Andrea 1994 {published data only}

D'Andrea M. The family development project: a comprehensive mental health counseling program for

pregnant adolescents. *Journal of Mental Health Counseling* 1994;**16**:184–95.

Elliott 2000 {published data only}

Elliott S, Leverton T, Sanjack M, Turner H, Cowmeadow P, Hopkins J, et al.Promoting mental health after childbirth: a controlled trial of primary prevention of postnatal depression. *British Journal of Clinical Psychology* 2000;**39**: 223–41.

Elliott SA, Sanjack M, Leverton TJ. Parents groups in pregnancy. A preventive intervention for postnatal depression?. In: Gottlieb BM editor(s). *Marshalling social support: formats, processes and effects.* Beverley Hills: Sage, 1988:87–97.

Gordon 1960 {published data only}

Gordon R, Gordon K. Social factors in prevention of postpartum emotional problems. *Obstetrics & Gynecology* 1960;**15**(4):433–8.

Gordon 1999 {published data only}

Gordon N, Walton D, McAdam E, Derman J, Gallitero G, Garrett L. Effects of providing hospital-based doulas in health maintenance organization hospitals. *Obstetrics & Gynecology* 1999;**93**(3):422–6.

Hayes 2001 {published data only}

Hayes B, Muller R, Bradley B. Perinatal depression: a randomized controlled trial of an antenatal education intervention for primiparas. *Birth* 2001;**28**(1):28–35.

Heh 2003 {published data only}

Heh S, Fu Y. Effectiveness of informational support in reducing the severity of postnatal depression. *Journal of Advanced Nursing* 2003;42:30–6.

Henderson 1998 {published data only}

Henderson J, Sharp J, Priest S, Hagan R, Evans S. Postnatal debriefing: what do women feel about it?. 4th Annual Congress of the Perinatal Society of Australia & New Zealand; 1998 March 30-April 4; Alice Springs, Australia. 1998:38.

Hodnett 2002 {published data only}

Hodnett E, Lowe N, Hannah M, Willan AR, Stevens B, Weston JA, et al.Effectiveness of nurses as providers of birth labor support in North American hospitals. *JAMA* 2002; **288**(11):1373–81.

Kealy 2003 {published data only}

Kealy M, Small R, Lumley J. Health and recovery after caesarean - is it as straightforward as we might want to believe?. Perinatal Society of Australia and New Zealand. 7th Annual Congress; 2003 March 9-12; Tasmania, Australia. 2003:A28.

Lieu 2000 {published data only}

Lieu T, Braveman P, Escobar G, Fischer A, Jensvold N, Capra A. A randomized comparison of home and clinic follow-up visits after early postpartum hospital discharge. *Pediatrics* 2000;**105**(5):1058–65.

Marks 2003 {published data only}

Marks MN, Siddle K, Warwick C. Can we prevent postnatal depression? A randomized controlled trial to assess the

effect of continuity of midwifery care on rates of postnatal depression in high-risk women. *Journal of Maternal-Fetal & Neonatal Medicine* 2003;**13**:119–27.

Oakley 1991 {published data only}

Oakley AR. Trial to reduce depression among mothers of young children - an intervention study. Personal communication 1991.

Okano 1998 {published data only}

Okano T, Nagata S, Hasegawa M, Nomura J, Kumar R. Effectiveness of antenatal education about postnatal depression: a comparison of two groups of Japanese mothers. *Journal of Mental Health* 1998;7(2):191–8.

Rees 1995 {published data only}

Rees BL. Effect of relaxation with guided imagery on anxiety, depression, and self-esteem in primiparas. *Journal* of *Holistic Nursing* 1995;**13**(3):255–67.

Saisto 2001 {published data only}

Saisto T, Salmela-Aro K, Nurmi J, Kononen T, Halmesmaki E. A randomized controlled trial of intervention in fear of childbirth. *Obstetrics & Gynecology* 2001;**98**(5 Pt 1):820–6.

Serwint 1991 {published data only}

Serwint J, Wilson M, Duggan A, Mellits E, Baumgardner R, DeAngelis C. Do postpartum nursery visits by the primary care provider make a difference?. *Pediatrics* 88;**3**:444–9.

Shields 1997 {published data only}

Shields N, Reid M, Cheyne H, Holmes A. Impact of midwife-managed care in the postnatal period: an exploration of psychosocial outcomes. *Journal of Reproductive & Infant Psychology* 1997;**15**(2):91–108.

Spinelli 1997 {published data only}

Spinelli, M. Interpersonal psychotherapy for depressed antepartum women: a pilot study. *American Journal of Psychiatry* 1997;**154**(7):1028–30.

Spinelli 2003 {published data only}

Spinelli MG, Endicott J. Controlled clinical trial of interpersonal psychotherapy versus parenting education program for depressed pregnant women. *American Journal* of *Psychiatry* 2003;**160**(3):555–62.

Stamp 1996 {published data only}

Stamp GE, Williams AS, Crowther CA. Predicting postnatal depression among pregnant women. *Birth* 1996;**23**: 218–23.

Webster 2003 {published data only}

Webster J, Linnane J, Roberts J, Starrenburg S, Hinson J, Dibley L. IDentify, Educate, and Alert (IDEA) trial: an intervention to reduce postnatal depression. *BJOG: an international journal of obstetrics and gynaecology* 2003;**110**: 842–6.

Wolman 1993 {published data only}

Nikodem C, Nolte A, Wolman W, Gulmezoglu A, Hofmeyr G. Companionship by a lay labour supporter to modify the clinical birth environment: long-term effects on mother and child. *Curationis* 1998;**21**(1):8–12. Wolman W, Chalmers B, Hofmeyr J, Nikodem C. Postpartum depression and companionship in the clinical

birth environment: a randomized controlled study. *American Journal of Obstetrics and Gynecology* 1993;**168**: 1388–93.

Zayas 2002 {published data only}

Zayas LH, Cunningham M, McKee MD, Jankowski KR. Depression and negative life events among pregnant africanamerican and hispanic women. *Womens Health Issues* 2002; **12**(1):16–22.

References to studies awaiting assessment

Blake 2001 {published data only}

Blake K. A randomised controlled trial to evaluate an antenatal and postnatal service by NEWPIN for disadvantaged mothers and their children. National Research Register (www.update-software.com/NRR) 2001 (accessed April 2004).

References to ongoing studies

Dennis 2004a *{unpublished data only}*

Dennis C-L, Hodnett E, Affonso D, Stewart DE, Zupancic J, Weston J. An RCT to evaluate the effect of peer (mother-to-mother) support for the prevention of postpartum depression among mothers at high-risk. Personal communication 2004.

Mann 2001 {published data only}

Mann A. A randomised control trial of a psychological intervention given in pregnancy to reduce the risk of postnatal depression in a sample of high risk women in India. National Research Register (www.updatesoftware.com/NRR) 2001 (accessed April 2004).

Additional references

Affonso 2000

Affonso DD, De AK, Horowitz JA, Mayberry LJ. An international study exploring levels of postpartum depressive symptomatology. *Journal of Psychosomatic Research* 2000;**49** (3):207–16.

Appleby 1994

Appleby L, Gregoire A, Platz C, Prince M, Kumar R. Screening women for high risk of postnatal depression. *Journal of Psychosomatic Research* 1994;**38**:539–45.

Appleby 1997

Appleby L, Warner R, Whitton A, Faragher B. A controlled study of fluoxetine and cognitive-behavioural counselling in the treatment of postnatal depression. *BMJ* 1997;**314**: 932–6.

Austin 2003

Austin M, Lumley J. Antenatal screening for postnatal depression: a systematic review. *Acta Psychiatrica Scandinavica* 2003;**107**:10–7.

Beck 1999

Beck CT. Maternal depression and child behaviour problems: a meta-analysis. *Journal of Advanced Nursing* 1999;**29**(3):623–9.

Beck 2001

Beck CT. Predictors of postpartum depression: an update. *Nursing Research* 2001;**50**(5):275–85.

Bernazzani 1997

Bernazzani O, Saucier J, David H, Borgeat F. Psychosocial predictors of depressive symptomatology level in postpartum women. *Journal of Affective Disorders* 1997;**46**:39–49.

Boyce 1994

Boyce P. Personality dysfunction, marital problems, and postnatal depression. In: Cox J, Holden J editor(s). *Perinatal psychiatry: use and misuse of the Edinburgh Postnatal Depression Scale*. London: Gaskell, 1994.

Breznitz 1988

Breznitz Z, Friedman SL. Toddlers' concentration: does maternal depression make a difference?. *Journal of Child Psychology & Psychiatry & Allied Disciplines* 1988;**29**(3): 267–79.

Brugha 1998

Brugha T, Sharp H, Cooper S, Weisender C, Britto D, Shinkwin R, et al. The Leister 500 Project: social support and the development of postnatal depressive symptoms, a prospective cohort survey. *Psychological Medicine* 1998;**28**: 63–79.

Buist 1998

Buist A. Child abuse, postpartum depression and parenting difficulties: a literature review of associations. *Australian and New Zealand Journal of Psychiatry* 1998;**32**:370–8.

Chen 1999

Chen CH, Wu HY, Tseng YF, Chou FH, Wang SY. Psychosocial aspects of Taiwanese postpartum depression phenomenological approach: a preliminary report. *Kao-Hsiung i Hsueh Ko Hsueh Tsa Chih [Kaohsiung Journal of Medical Sciences]* 1999;**15**:44–51.

Cogill 1986

Cogill S, Caplan H, Alexandra H, Robson K, Kumar R. Impact of postnatal depression on cognitive development of young children. *BMJ* 1986;**292**:1165–7.

Cohn 1983

Cohn JF, Tronick EZ. Three-month-old infants' reaction to simulated maternal depression. *Child Development* 1983;**54** (1):185–93.

Cooper 1995

Cooper P, Murray L. The course and recurrence of postnatal depression. *British Journal of Psychiatry* 1995;**166**:191–5.

Cooper 1996

Cooper P, Murray L, Hooper R, West A. The development and validation of a predictive index for postpartum depression. *Psychological Medicine* 1996;**26**:627–34.

Cooper 1997

Cooper P, Murray L. The impact of psychological treatments of postpartum depression on maternal mood and infant development. In: Cooper P, Murray L editor(s). *Postpartum depression and child development*. New York: Guilford, 1997:201–20.

Cooper 1998

Cooper P, Murray L. Postnatal depression. *BMJ* 1998;**316**: 1884–6.

Cooper 2003

Cooper PJ, Murray L, Wilson A, Romaniuk H. Controlled trial of the short- and long-term effect of psychological treatment of post-partum depression. I. Impact on maternal mood. *British Journal of Psychiatry* 2003;**182**:412–419.

Cox 1987

Cox AD, Puckering C, Pound A, Mills M. The impact of maternal depression in young children. *Journal of Child Psychology & Psychiatry & Allied Disciplines* 1987;**28**(6): 917–28.

Cox 1993

Cox J, Murray D, Chapman G. A controlled study of the onset, duration and prevalence of postnatal depression. *British Journal of Psychiatry* 1993;**163**:27–31.

Cummings 1994

Cummings EM, Davies PT. Maternal depression and child development. *Journal of Child Psychology & Psychiatry & Allied Disciplines* 1994;**35**(1):73–112.

Dennis 2003

Dennis C-L. The effect of peer support on postpartum depression: a pilot randomized controlled trial. *Canadian Journal of Psychiatry* 2003;**48**:61–70.

Dennis 2004b

Dennis C-L. Can we identify mothers at-risk for postpartum depression in the immediate postpartum period using the Edinburgh Postnatal Depression Scale?. *Journal of Affective Disorders* 2004;78:163–169.

Evins 1997

Evins G, Theofrastous J. Postpartum depression: a review of postpartum screening. *Primary Care Update for Ob/Gyns* 1997;**4**(6):241–6.

Hannah 1992

Hannah P, Adams D, Lee A, Glover V, Sandler M. Links between early post-partum mood and post-natal depression. *British Journal of Psychiatry* 1992;**160**:777–80.

Higgins 2002

Higgins J, Thompson SG. Quantifying heterogeneity in a meta-analysis. *Statistics in Medicine* 2002;**21**:1539–58.

Hipwell 2000

Hipwell AE, Goossens FA, Melhuish EC, Kumar R. Severe maternal psychopathology and infant-mother attachment. *Developmental Psychopathology* 2000;**12**(2):157–75.

Holden 1989

Holden J, Sagovsky R, Cox J. Counselling in a general practice setting: controlled study of health visitor intervention in treatment of postnatal depression. *BMJ* 1989;**298**:223–6.

Holden 1991

Holden JM. Postnatal depression: its nature, effects, and identification using the Edinburgh Postnatal Depression scale. *Birth* 1991;**18**(4):211–21.

Lane 1997

Lane A, Kevill R, Morris M, Turner M, Barry S. Postnatal depression and elation among mothers and their partners: prevalence and predictors. *British Journal of Psychiatry* 1997; **171**:550–5.

Mayberry 1993

Mayberry L, Affonso D. Infant temperament and postpartum depression: a review. *Health Care for Women International* 1993;14:206–11.

McLennan 2002

McLennan JD, Offord DR. Should postpartum depression be targeted to improve child mental health?. *Journal of the American Academy of Child & Adolescent Psychiatry* 2002;**41**: 28–35.

Mrazek 1994

Mrazek PJ, Haggerty RJ. *Reducing risks for mental disorders* - frontiers for prevention intervention research. Washington, D.C: National Academy Press, 1994.

Murray 1992

Murray L. The impact of postnatal depression on child development. *Journal of Child Psychology and Psychiatry* 1992;**33**:543–61.

Murray 1996

Murray L, Fiori-Cowley A, Hooper R, Cooper P. The impact of postnatal depression and associated adversity on early mother-infant interactions and later infant outcome. *Child Development* 1996;**67**(5):2512–26.

Murray 1999

Murray L, Sinclair D, Cooper P, Ducournau P, Turner P, Stein A. The socioemotional development of 5-year-old children of postnatally depressed mothers. *Journal of Child Psychology & Psychiatry & Allied Disciplines* 1999;**40**(8): 1259–71.

Nahas 1999

Nahas VL, Hillege S, Amasheh N. Postpartum depression: the lived experiences of Middle Eastern migrant women in Australia. *Journal of Nurse-Midwifery* 1999;44:65–74.

O'Hara 1986

O'Hara M. Social support, life events, and depression during pregnancy and the puerperium. *Archives of General Psychiatry* 1986;**43**:569–73.

O'Hara 1991

O'Hara M, Schlecte J, Lewis D, Wright E. Prospective study of postpartum blues: biologic and psychosocial factors. *Archives of General Psychiatry* 1991;**48**:801–6.

O'Hara 1996

O'Hara M, Swain A. Rates and risk of postpartum depression - a meta-analysis. *International Review of Psychiatry* 1996;**8**:37–54.

O'Hara 1997

O'Hara M. The nature of postpartum depressive disorders. In: Murray L, Cooper P editor(s). *Postpartum depression and child development*. New York: Guilford, 1997:3–31.

O'Hara 2000

O'Hara MW, Stuart S, Gorman LL, Wenzel A. Efficacy of interpersonal psychotherapy for postpartum depression. *Archives of General Psychiatry* 2000;**57**:1039–45.

Orvaschel 1988

Orvaschel H, Walsh-Allis G, Ye WJ. Psychopathology in children of parents with recurrent depression. *Journal of Abnormal Child Psychology* 1988;**16**(1):17–28.

RevMan 2000

The Cochrane Collaboration. Review Manager (RevMan). 4.1 for Windows. Oxford, England: The Cochrane Collaboration, 2000.

Ritter 2000

Ritter C, Hobfoll SE, Lavin J, Cameron RP, Hulsizer MR. Stress, psychosocial resources, and depressive symptomatology during pregnancy in low-income, innercity women. *Health Psychology* 2000;**19**(6):576–85.

Shah 1998

Shah CP. *Public health and preventive medicine in Canada*. Fourth. Toronto: University of Toronto Press, 1998.

Small 1994

Small R, Astbury J, Brown S, Lumley J. Depression after childbirth: does social context matter?. *Medical Journal of Australia* 1994;**161**:473–7.

Stein 1989

Stein A, Cooper P, Campbell E. Social adversity and perinatal complications: their relation to postnatal depression. *BMJ* 1989;**298**:1073–4.

Stocky 2000

Stocky A, Lynch J. Acute psychiatric disturbance in pregnancy and the puerperium. *Baillieres Best Practice Research Clinical Obstetrics and Gynaecology* 2000;**14**(1): 73–87.

Stuchbery 1998

Stuchbery M, Matthey S, Barnett B. Postnatal depression and social support in Vietnamese, Arabic and Anglo-Celtic mothers. *Social Psychiatry and Psychiatric Epidemiology* 1998;**33**:483–90.

Whiffen 1989

Whiffen VE, Gotlib IH. Infants of postpartum depressed mothers: temperament and cognitive status. *Journal of Abnormal Psychology* 1989;**98**(3):274–9.

Wickberg 1996

Wickberg B, Hwang C. Counselling of postnatal depression: a controlled study on a population based Swedish sample. *Journal of Affective Disorders* 1996;**39**:209–16.

Will 2000

Will TA, Shinar O. Measuring perceived and received social support. In: Cohen S, Underwood L, Gottlieb B editor (s). Social support measurement and intervention: a guide for health and social scientists. Toronto: Oxford University Press, 2000.

Yamashita 2000

Yamashita H, Yoshida K, Nakano H, Tashiro N. Postnatal depression in Japanese women: detecting the early onset of

postnatal depression by closely monitoring the postpartum mood. *Journal of Affective Disorders* 2000;**58**:145–54. * *Indicates the major publication for the study*

CHARACTERISTICS OF STUDIES

Characteristics of included studies [ordered by study ID]

Armstrong 1999

Methods	RCT - randomisation was performed using a computer generated random numbers table and completed by clerical staff not involved in the eligibility assessment. A power analysis was performed and the outcome assessor was blinded to group allocation. Nurses providing the intervention were also blinded to 6 weeks postpartum (within usual care parameters). The 16-week attrition rate was 12%.		
Participants	181 mothers (90 in the intervention group; 91 in the control group) who gave birth in one urban hospital in Queensland, Australia. Families were included where the child, for environmental reasons, was at increased risk for poor health and developmental outcomes. Exclusion criteria included poor English literacy skills		
Interventions	Intervention group: weekly nursing home visits for the first 6 weeks, fortnightly until 12 weeks, then monthly until 24 weeks. Mothers were also encouraged to access existing community services. Control group: standard care which included encouragement to access existing community services, an offer for home visits by a nurse (usually limited to 1 visit), and no limit on number of centre visits (by appointment only)		
Outcomes	Outcomes included depression (EPDS > 12), parental stress (Parenting Stress Index), breastfeeding dura- tion, infant immunisation and utilisation of medical services and, accidental injury at 6 and 16 weeks		
Notes	significantly more primiparous and aboriginal mo	ning questionnaire. The intervention group included others and fewer women (1) with a past history of sychiatric illness, and (3) who reported physical forms	
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Yes	Yes A - Adequate	
Brugha 2000			
Methods	RCT - randomisation was performed using a compu	Iter-based stratification process with minimisation on	
	three prognostic factors (level of support, screening and outcome assessors were blinded to group alloca	, and ethnic group). A power analysis was performed tion. The 12-week attrition rate was 9%	

Brugha 2000 (Continued)

Interventions	Intervention group: 'Preparing for Parenthood' - 6 structured 2-hour weekly antenatal classes (preceded by an initial introductory meeting with the participant and her partner) and 1 'reunion' class at 8 weeks postpartum. Classes were provided by a trained nurse and occupational therapist and based on established psychological models for tackling depression together with emerging models for enhancing social support	
Outcomes	Outcomes included depression (EPDS > 10) and maternal health service contact since randomization at 12 weeks postpartum	
Notes	Women in the intervention group were more likely to adopt an avoidant problem-solving style than women in the control group; using logistic modelling to adjust for this covariate at baseline did not alter the trial results. Only 45% of participants in the intervention group attended sufficient sessions to 'likely receive benefit'	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate
Gamble 2003 Methods		tively numbered, sealed, opaque envelopes. A power is blinded to group allocation. The 12-week attrition
Participants	103 mothers (50 in the intervention group; 53 in the control group) who were assessed for labour trauma risk in the immediate postpartum period in a Brisbane, Australia hospital	
Interventions	Intervention group: 1 midwifery-led debriefing session before hospital discharge and another at 6 to 8 weeks postpartum. Control group: standard care with no midwifery-led debriefing session	
Outcomes	Outcomes included depression (EPDS > 12) at 12 weeks postpartum	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate

Gorman	2002

Methods	RCT - randomisation was performed using a random numbers table and a blocking strategy based on the	
	presence or absence of current or past history of depression. Outcome data were collected via interview and mailed questionnaires. The 24-week attrition rate was 18%	
Participants	45 pregnant women (24 in the intervention group; 21 in the control group) at-risk for postpartum depression who attended various obstetric clinics in Iowa City and St. Louis, USA	
Interventions	Intervention group: 5 individual sessions based on interpersonal psychotherapy, beginning in late preg- nancy and ending at approximately 4 weeks postpartum	
Outcomes	Outcomes included depression (EPDS > 12 and SCID) at 4 and 24 weeks postpartum	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate
Gunn 1998		
Gunn 1998 Methods	RCT - randomisation was performed via telephone	through a centrally controlled randomisation centre.
	A power analysis was conducted and outcome data w attrition rate was 34% 683 healthy mothers (number of women randomis	rere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one
Methods	A power analysis was conducted and outcome data w attrition rate was 34% 683 healthy mothers (number of women randomis rural and one metropolitan hospital in Victoria, Aus	rere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one stralia. Women were excluded if they were patients of
Methods	A power analysis was conducted and outcome data w attrition rate was 34% 683 healthy mothers (number of women randomis rural and one metropolitan hospital in Victoria, Aus general practitioners who were the trial reference gr emergency caesarean section	through a centrally controlled randomisation centre. vere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one stralia. Women were excluded if they were patients of roup, attended the teenage clinic, or delivered by an date to see a general practitioner for a check-up: the ge and the control group for 6 weeks postpartum
Methods Participants	A power analysis was conducted and outcome data w attrition rate was 34% 683 healthy mothers (number of women randomis rural and one metropolitan hospital in Victoria, Aus general practitioners who were the trial reference gr emergency caesarean section All participants received a letter and appointment of intervention group for 1 week after hospital discharg	rere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one stralia. Women were excluded if they were patients of roup, attended the teenage clinic, or delivered by an date to see a general practitioner for a check-up: the ge and the control group for 6 weeks postpartum
Methods Participants Interventions	A power analysis was conducted and outcome data wattrition rate was 34% 683 healthy mothers (number of women randomis rural and one metropolitan hospital in Victoria, Aus general practitioners who were the trial reference gr emergency caesarean section All participants received a letter and appointment of intervention group for 1 week after hospital discharg Outcomes included depression (EPDS > 12), mat	rere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one stralia. Women were excluded if they were patients of roup, attended the teenage clinic, or delivered by an date to see a general practitioner for a check-up: the ge and the control group for 6 weeks postpartum
Participants Interventions Outcomes	A power analysis was conducted and outcome data wattrition rate was 34% 683 healthy mothers (number of women randomis rural and one metropolitan hospital in Victoria, Aus general practitioners who were the trial reference gr emergency caesarean section All participants received a letter and appointment of intervention group for 1 week after hospital discharg Outcomes included depression (EPDS > 12), mat	rere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one stralia. Women were excluded if they were patients of roup, attended the teenage clinic, or delivered by an date to see a general practitioner for a check-up: the
Methods Participants Interventions Outcomes Notes	A power analysis was conducted and outcome data wattrition rate was 34% 683 healthy mothers (number of women randomis rural and one metropolitan hospital in Victoria, Aus general practitioners who were the trial reference gr emergency caesarean section All participants received a letter and appointment of intervention group for 1 week after hospital discharg Outcomes included depression (EPDS > 12), mat	rere collected via mailed questionnaires. The 24-week ed to each group not stated) who gave birth in one stralia. Women were excluded if they were patients of roup, attended the teenage clinic, or delivered by an date to see a general practitioner for a check-up: the ge and the control group for 6 weeks postpartum

Lavender 1998		
Methods	RCT - randomisation was performed using computer-generated numbers and by opening consecutively numbered, sealed opaque envelopes. A power analysis was conducted and outcome data were collected via a mailed questionnaire. The 3-week attrition rate was 5%	
Participants	114 primiparous mothers (60 in the intervention group; 60 in the control group) in a UK teaching hospital. Inclusion criteria: singleton pregnancy, cephalic presentation, spontaneous labour at term, normal vaginal delivery	
Interventions	Intervention group: 1 debriefing session before hospital discharge, which lasted 30 to 120 minutes, provided by a midwife who received no formal training. Control group: standard care with no midwifery-led debriefing session	
Outcomes	Outcomes included depression and anxiety (Hospital Anxiety Depression Scale - HAD > 10) at 3 weeks postpartum	
Notes	Atypical population - 59.6% were single mothers.	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate

MacArthur 2002

Methods	independent clinical trials unit. 17 practices were raw were randomised to the control group. A power and	RCT with cluster design - randomisation was performed using a customised, computer program in an independent clinical trials unit. 17 practices were randomised to the intervention group and 19 practices were randomised to the control group. A power analysis was conducted and outcome data were collected via mailed questionnaires. The 16-week attrition rate was 27%	
Participants		2064 UK mothers (1087 in the intervention group; 977 in the control group). Only mothers expected to move out of the general practice area were excluded	
Interventions	included (1) screening with a symptoms checklist a necessary, and (3) a 10-12 week discharge visit. Co home visits to 10-14 days postpartum (may exter	Intervention group: flexible, individualised, extended home visits by a midwife to 28 days postpartum that included (1) screening with a symptoms checklist and the EPDS, (2) a referral to a general practitioner as necessary, and (3) a 10-12 week discharge visit. Control group: standard care that included 7 midwifery home visits to 10-14 days postpartum (may extend to 28 days) and care by health visitors thereafter. General practitioners completed routine home visits and a final check-up at 6 to 8 weeks postpartum	
Outcomes	Outcomes included depression (EPDS > 12) at 16	Outcomes included depression (EPDS > 12) at 16 weeks postpartum	
Notes			
Risk of bias			
Item	Authors' judgement	Description	

MacArthur 2002 (Continued)

Allocation concealment?	Yes	A - Adequate
Morrell 2000		
Methods		ing a random numbers table and by opening consecutively num- rer analysis was conducted and outcome data were collected via
Participants	623 UK mothers (311 in the intervention group; 312 in the control group). Exclusion criteria: insufficient English to complete questionnaires and an infant in the special care unit for more than 48 hours	
Interventions	Intervention group: postnatal care at home by community midwives plus up to 10 home visits in the first month postpartum lasting up to 3 hours provided by a community postnatal support worker. Control group: postnatal care at home by community midwives	
Outcomes	Outcomes included depression (EPDS > 12), maternal physical and mental well-being (SF-36), social support (Duke Functional Social Support), and breastfeeding duration at 6 and 24 weeks postpartum	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate
Priest 2003		
Methods	an envelope from a group of at least six	hin the strata of parity and mode of delivery. Each woman selected sealed, opaque envelopes containing random allocation. A power ta were collected via mailed questionnaires. The 52-week attrition
Participants	1745 Australian mothers (875 in the intervention group; 870 in the control group). Exclusion criteria: insufficient English to complete questionnaires, being under psychological care at the time of delivery, maternal age < 18 years, and infant needing neonatal intensive care	
Interventions	Intervention group: a single, standardised debriefing session provided in-hospital immediately after ran- domisation or the next day; duration ranged from 15 minutes to 1 hour and all research midwives received training in critical incident stress debriefing. Control group: standard postpartum care.	
Outcomes	Outcomes included depression (EPDS	> 12) at 8, 24, and 52 weeks postpartum

Priest 2003 (Continued)

Risk of bias		
Item	Authors' judgement Description	
Allocation concealment?	Yes	A - Adequate
Reid 2002		
Methods	RCT with a 2 x 2 factorial design - randomisation was performed using a computer generated scheme with randomised permuted blocks, stratified by centre. A power analysis was conducted and outcome data were collected via mailed questionnaires. The 24-week attrition rate was 29%	
Participants	1004 UK mothers (503 in the intervention group; 501 in the control group). Inclusion criteria: all primiparous women attending antenatal clinics in two participating hospitals. Exclusion criteria: women whose infant subsequently died or was admitted to the Special Care Unit for more than 2 weeks	
Interventions	Two postpartum interventions incorporating 4 groups: control, mailed self-help materials, invitation to support group, and self-help materials plus invitation to support group. Data was analysed by pooling the four groups as self-help vs no self-help and support group vs no support group. The support groups were run on a weekly basis for 2-hours facilitated by trained midwives	
Outcomes	Outcomes included depression (EPDS > 11), maternal physical and mental well-being (SF-36), and social support (SSQ6) at 12 and 24 weeks postpartum	
Notes	For this review, only the support group vs no support group comparisons were included. Only 18% or participants in the intervention group attended a support group session	
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes	A - Adequate
Small 2000		
Methods	RCT - randomisation was performed via telephone using a computer generated randomisation schedule for each midwife. A power analysis was conducted and outcome data were collected via a mailed questionnaire. The 24-week attrition rate was 12%	
Participants	1041 mothers (520 in the intervention group; 521 in the control group) who had an operative delivery in a large maternity teaching hospital in Melbourne, Australia	
Interventions	Intervention group: a midwifery-led debriefing session before discharge to provide women with an op- portunity to discuss their labour, birth, and postdelivery events and experiences. Control group: standard care which included a brief visit from a midwife on discharge to give a pamphlet on sources of assistance	

Small 2000 (Continued)

Outcomes	Outcomes included depression (EPDS > 12) and overall maternal health status (SF-36) at 24-weeks postpartum	
Notes		
Risk of bias		
Item	Authors' judgement	Description
Allocation concealment?	Yes A - Adequate	
Stamp 1995		
Methods		utively numbered, sealed opaque envelopes with strat- l and outcome data were collected via a mailed ques-
Participants	144 pregnant women (73 in the intervention group; 71 in the control group) who screened at-risk for postpartum depression during antenatal clinic visits in Adelaide, Australia. Inclusion criteria: English-speaking, singleton fetus, and < 24 weeks gestation	
Interventions	Intervention group: routine antenatal care plus 2 antenatal and 1 postnatal midwifery-led group sessions. Control group: routine antenatal and postnatal care which included a class at 6 weeks postpartum that incorporated a video on postpartum depression	
Outcomes	Outcomes included depression (EPDS > 12) at 6, 12, and 24 weeks postpartum	
Notes	A high number of women screened 'vulnerable' and attended all 3 sessions	d only 31% of participants in the intervention group
Risk of bias		
Item	Authors' judgement Description	
Allocation concealment?	Yes A - Adequate	
Tam 2003		
Methods		m numbers table and consecutively numbered, sealed, d and outcome data were collected via interview and
Participants	560 in-hospital Chinese mothers (280 in each group) with at least one suboptimal outcome in the perinatal period ranging from antenatal complications requiring hospitalisation, elective caesarean section, labour induction, postpartum haemorrhage, infant admission to special care unit, etc	

Tam 2003 (Continued)

Interventions	Intervention group: routine postpartum care plus 1 to 4 sessions of "educational counselling" by a research nurse before hospital discharge that included information related to the adverse event and counselling to assist the mother to "come to terms with her losses and find solutions to specific difficulties" (median total time of was 35 minutes). Twenty-four women also received one session by a physician		
Outcomes	Outcomes included depression (HADS > 4) at 6 weeks postpartum		
Notes	Health professionals were not blinded to group allocation.		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Yes	A - Adequate	
Waldenstrom 2000			
Methods	RCT - randomisation was performed via telephone using consecutively numbered, sealed, opaque envelopes. A power analysis was conducted and outcome data were collected via mailed questionnaires. The 8-week attrition rate was 7%		
Participants	1000 pregnant low-risk mothers (495 in the intervention group; 505 in the control group) attending an antenatal clinic in Melbourne, Australia. Inclusion criteria: > 25 weeks gestation, English-speaking, and low medical risk		
Interventions	Intervention group: team midwifery care provided antenatally and postnatally in hospital with a focus on continuity. Control group: standard antenatal and postnatal care by physicians and midwives with no focus on continuity		
Outcomes	Outcomes included depression (EPDS > 12) at 8 weeks postpartum		
Notes	The primary outcome of this study was satisfaction with care. 1000 women were randomised with 83 unavoidable exclusions (intervention group = 39; control group = 44). Demographic differences were found between questionnaire responders and non-responders		
Risk of bias			
Item	Authors' judgement	Description	
Allocation concealment?	Yes	A - Adequate	

Zlotnick	2001
LIUUIICA	2001

Methods	RCT - unclear randomisation process. Outcome data were collected via interview. The 12-week attrition rate was 6%					
Participants	37 pregnant women (17 in the intervention group; 18 in the control group) on public assistance who had at least 1 risk factor for postpartum depression and were attending a prenatal clinic at a general hospital in the northeast USA					
Interventions	Intervention group: "Survival Skills for New Moms", which involved four 60-minute group sessions over a 4-week period based on the principles of interpersonal psychotherapy. Control group: standard antenatal care					
Outcomes	Outcomes included depression (SCID) at 12 weeks postpartum.					
Notes	Notes 50% of eligible women declined trial participation. Atypical sample as 77% of participants were single women. It is unknown who collected follow-up data or provided the intervention					
Risk of bias						
Item	Authors' judgement Description					
Allocation concealment?	Unclear B - Unclear					

RCT: Randomised controlled trial EPDS: Edinburgh Postnatal Depression Scale vs: versus SCID: Structured Clinical Interview for DSM-IV

Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Buist 1999	Pilot trial with unclear randomisation method. Significant group differences in baseline characteristics. No usable outcome data; published data were mean scores without standard deviations
Chabrol 2002	Not an RCT. Odd versus even number group assignment was used. Data were not analysed using 'intent-to-treat'
Cooper 2002	Not an RCT. Study examined the impact of a mother-infant intervention through the comparison between two matched groups
D'Andrea 1994	Postpartum depression was not a study outcome.
Elliott 2000	Not an RCT. Group allocation based on delivery date. Potential selection bias with significant differences between participating and non-participating eligible women. Data were presented using median instead of mean results

(Continued)

Gordon 1960	Not an RCT. Inexplicit non-random group allocation. Primary outcome was 'emotional upset' using a subjective measure. All participant characteristics were lacking and 46% of mothers were lost to follow up
Gordon 1999	A poor measure of postpartum depression was used that included a single item question and subscore on the mental health index of the SF-36. In addition, 30% women were excluded postrandomisation
Hayes 2001	Intervention was not psychosocial or psychological, but rather included a single educational session about post- partum depression, provided antenatally by a midwife
Heh 2003	Intervention was not psychosocial or psychological but rather included only information related to postpartum depression
Henderson 1998	Not an RCT but examines data that were part of the Priest 2003 trial
Hodnett 2002	The intervention (continuous intrapartum support) was neither psychological nor psychosocial. Postpartum depression was not the primary or secondary outcome
Kealy 2003	Not an RCT.
Lieu 2000	Premature assessment of postpartum depression (2 weeks after delivery), which was neither the primary nor secondary outcome
Marks 2003	Approximately 25% of participants were currently suffering from depression at recruitment and 49% had a depressive episode sometime during the perinatal period
Oakley 1991	Intervention was not targeting the prevention of postpartum depression but depression among mothers of young children
Okano 1998	Not an RCT. Study examined an educational session retrospectively involving two non-randomised groups of women who sought psychiatric care postnatally
Rees 1995	Intervention was not targeting the prevention of postpartum depression but rather the treatment of depression among pregnant women
Saisto 2001	Postpartum depression was neither a primary or secondary outcome; statistical results related to postpartum depression were not reported
Serwint 1991	Not an RCT. Group allocation was based on a 2-week period.
Shields 1997	Study reports on an element of a larger trial where the primary and secondary outcome was not postpartum depression. Furthermore, one EPDS item (self-harm) was excluded rendering the clinical interpretability of the outcome data questionable
Spinelli 1997	Not an RCT. A single-group study evaluating an interpersonal psychotherapy intervention for the treatment of antepartum depression

(Continued)

Spinelli 2003	Intervention was not targeting the prevention of postpartum depression but rather the treatment of antepartum depression
Stamp 1996	Not an RCT.
Webster 2003	The intervention in this well-conducted trial was not psychosocial or psychological but rather included antenatal identification as high-risk, an educational booklet and discussion about the risk of developing postpartum depression, and a letter to the woman's referring general practitioner and local Child Health Nurse alerting them of the woman's risk
Wolman 1993	The researchers significantly changed the study protocol before trial completion. Inability to assess selection bias. Trial had a 21% loss to follow up and a poor measure of postpartum depression (Pitt Depression Inventory) was used for the main portion of the trial
Zayas 2002	While the author identified the study as an RCT, no information was provided related to the randomisation process or the intervention. It is also unknown whether the outcome assessor was blinded or whether the data were analysed using 'intent-to-treat'

EPDS: Edinburgh Postnatal Depression Scale RCT: Randomised controlled trial

Characteristics of ongoing studies [ordered by study ID]

Dennis 2004a

Trial name or title	A randomised controlled trial to evaluate the effect of peer (mother-to-mother) support for the prevention of postpartum depression among high-risk mothers
Methods	
Participants	700 Canadian mothers who had an EPDS score > 9 within 24-48 hours following hospital discharge. No previous history of psychosis and currently not taking anti-depressant medications
Interventions	Telephone-based peer support provided by an experienced mother who previously suffered and recovered from postpartum depression and participated in a 4-hour training session
Outcomes	Postpartum depression at 12 and 24 weeks postpartum as measured by a clinical interview (SCID) and EPDS
Starting date	Funding started April 2004.
Contact information	Dr. Cindy-Lee Dennis Assistant Professor Faculty of Nursing University of Toronto Email: cindylee.dennis@utoronto.ca

Dennis 2004a (Continued)

Notes	
Mann 2001	
Trial name or title	A randomised controlled trial of a psychological intervention given in pregnancy to reduce the risk of postnatal depression in a sample of high risk women in India
Methods	
Participants	423 pregnant Indian women identified as high-risk based on a researcher developed risk score.
Interventions	Home-based 'listening visits' provided from 30 weeks gestation to 10 weeks postpartum
Outcomes	Postpartum depression at 6, 12, and 24 weeks as measured using the EPDS and a revised clinical interview schedule providing a diagnosis according to ICD-10 criteria
Starting date	Data collection to end June 2004.
Contact information	Dr. Anthony Mann Institute of Psychiatry De Crespigny Park Denmark Hill London, UK Email: spjuahm@iop.kcl.ac.uk
Notes	Trial information was provided by : Dr. Marcus Hughes Wellcome Trust Research Fellow Institute of Psychiatry De Crespigny Park Denmark Hill London, UK Email: m.hughes@iop.kcl.ac.uk

EPDS: Edinburgh Postnatal Depression Scale

ICD: International Classification of Diseases

SCID: Structured clinical interview for the diagnostic and statistical manual of mental disorders

DATA AND ANALYSES

Comparison 1. All interventions versus usual care - various study outcomes

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Depressive symptomatology at final study assessment (variously defined)	15		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
1.1 Depressive symptomatology - all trials	15	7697	Risk Ratio (M-H, Random, 95% CI)	0.81 [0.65, 1.02]
1.2 Sensitivity analysis	9	3978	Risk Ratio (M-H, Random, 95% CI)	0.76 [0.52, 1.12]
2 Depressive symptomatology at final study assessment (EPDS > 12)	10		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
2.1 Depressive symptomatology - all trials	10	6126	Risk Ratio (M-H, Random, 95% CI)	0.91 [0.73, 1.15]
2.2 Sensitivity analysis	7	3667	Risk Ratio (M-H, Random, 95% CI)	1.00 [0.75, 1.34]
3 Mean depression scores at final study assessment	8		Mean Difference (IV, Random, 95% CI)	Subtotals only
3.1 Mean depression scores - all trials	8	4880	Mean Difference (IV, Random, 95% CI)	-0.36 [-1.21, 0.48]
3.2 Sensitivity analysis	3	1107	Mean Difference (IV, Random, 95% CI)	0.23 [-0.43, 0.89]
4 Depressive symptomatology at 8, 16, 24 weeks postpartum (variously defined)	15		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
4.1 Depressive symptomatology at 0 to 8 weeks - all trials	8	4091	Risk Ratio (M-H, Random, 95% CI)	0.65 [0.43, 1.00]
4.2 Depressive symptomatology at 0 to 8 weeks - sensitivity analysis	7	3575	Risk Ratio (M-H, Random, 95% CI)	0.62 [0.38, 1.03]
4.3 Depressive symptomatology at 9 to 16 weeks - all trials	8	3326	Risk Ratio (M-H, Random, 95% CI)	0.80 [0.56, 1.12]
4.4 Depressive symptomatology at 9 to 16 weeks - sensitivity analysis	6	1788	Risk Ratio (M-H, Random, 95% CI)	0.87 [0.59, 1.28]
4.5 Depressive symptomatology at 17 to 24 weeks	7	4314	Risk Ratio (M-H, Random, 95% CI)	1.02 [0.87, 1.19]
5 Depressive symptomatology at 8, 16, 24 weeks postpartum (EPDS > 12)	10		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
5.1 Depressive symptomatology at 0 to 8 weeks	6	3452	Risk Ratio (M-H, Random, 95% CI)	0.90 [0.65, 1.25]

5.2 Depressive symptomatology at 9 to 16	5	2369	Risk Ratio (M-H, Random, 95% CI)	0.72 [0.49, 1.06]
weeks - all trials 5.3 Depressive symptomatology at 9 to 16	4	866	Risk Ratio (M-H, Random, 95% CI)	0.70 [0.39, 1.28]
weeks - sensitivity analysis 5.4 Depressive symptomatology at 17 to 24 weeks	6	3598	Risk Ratio (M-H, Random, 95% CI)	1.00 [0.84, 1.19]
6 Maternal health service contact at final study assessment	1	190	Risk Ratio (M-H, Fixed, 95% CI)	1.21 [0.57, 2.56]
6.1 Maternal health service contact	1	190	Risk Ratio (M-H, Fixed, 95% CI)	1.21 [0.57, 2.56]
7 Maternal-infant attachment at 8, 16, 24 weeks postpartum	1		Mean Difference (IV, Fixed, 95% CI)	Subtotals only
7.1 Maternal-infant attachment at 0 to 8 weeks	1	174	Mean Difference (IV, Fixed, 95% CI)	-0.10 [-1.18, 0.98]
7.2 Maternal-infant attachment at 9 to 16 weeks	1	160	Mean Difference (IV, Fixed, 95% CI)	-0.73 [-1.84, 0.38]
8 Maternal attitudes toward motherhood at 8, 16, 24 weeks postpartum	1		Mean Difference (IV, Fixed, 95% CI)	Subtotals only
8.1 Restriction of role at 0 to 8 weeks	1	174	Mean Difference (IV, Fixed, 95% CI)	-0.13 [-1.88, 1.62]
8.2 Restriction of role at 9 to 16 weeks	1	160	Mean Difference (IV, Fixed, 95% CI)	-1.12 [-2.95, 0.71]
9 Maternal anxiety at 8, 16, 24 weeks postpartum	1		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
9.1 Maternal anxiety at 0 to 8 weeks	1	114	Risk Ratio (M-H, Fixed, 95% CI)	0.14 [0.05, 0.37]
10 Competence in mothering at 8 and 16 weeks postpartum	1		Mean Difference (IV, Fixed, 95% CI)	Subtotals only
10.1 Maternal competence at 0 to 8 weeks	1	174	Mean Difference (IV, Fixed, 95% CI)	0.34 [-1.68, 2.36]
10.2 Maternal competence at 9 to 16 weeks	1	160	Mean Difference (IV, Fixed, 95% CI)	-1.46 [-3.84, 0.92]
11 General physical and mental health (SF-36) at final study assessment	4		Mean Difference (IV, Fixed, 95% CI)	Subtotals only
11.1 Physical functioning	4	2589	Mean Difference (IV, Fixed, 95% CI)	0.29 [-0.91, 1.49]
11.2 Role functioning (physical)	4	2588	Mean Difference (IV, Fixed, 95% CI)	-0.90 [-3.33, 1.52]
11.3 Bodily pain	4	2589	Mean Difference (IV, Fixed, 95% CI)	0.25 [-1.41, 1.92]
11.4 Mental health	4	2582	Mean Difference (IV, Fixed, 95% CI)	-0.85 [-2.21, 0.52]
11.5 Role functioning (emotional)	4	2586	Mean Difference (IV, Fixed, 95% CI)	-0.93 [-3.55, 1.69]
11.6 Vitality	4	2581	Mean Difference (IV, Fixed, 95% CI)	0.64 [-0.99, 2.28]
11.7 Social functioning	4	2591	Mean Difference (IV, Fixed, 95% CI)	-0.59 [-2.29, 1.10]
11.8 General health	4	2586	Mean Difference (IV, Fixed, 95% CI)	-0.19 [-1.68, 1.29]
12 Perceived social support at 8, 16, 24 weeks postpartum	2		Std. Mean Difference (IV, Fixed, 95% CI)	Subtotals only

12.1 Perceived social support at 0 to 8 weeks	1	513	Std. Mean Difference (IV, Fixed, 95% CI)	0.01 [-0.16, 0.19]
12.2 Perceived social support at 9 to 16 weeks	1	732	Std. Mean Difference (IV, Fixed, 95% CI)	Not estimable
12.3 Perceived social support at 17 to 24 weeks	2	1174	Std. Mean Difference (IV, Fixed, 95% CI)	0.02 [-0.09, 0.14]
13 Breastfeeding duration at 8, 16, 24 weeks postpartum	3		Risk Ratio (M-H, Fixed, 95% CI)	Subtotals only
13.1 Breastfeeding at 0 to 8 weeks	2	722	Risk Ratio (M-H, Fixed, 95% CI)	1.03 [0.89, 1.19]
13.2 Breastfeeding at 9 to 16 weeks	2	635	Risk Ratio (M-H, Fixed, 95% CI)	0.89 [0.76, 1.04]
13.3 Breastfeeding at 17 to 24 weeks	2	968	Risk Ratio (M-H, Fixed, 95% CI)	0.90 [0.74, 1.10]
14 Infant immunisations	1	160	Mean Difference (IV, Fixed, 95% CI)	0.42 [0.11, 0.73]
14.1 Mean number of immunizations at 12 to 16 weeks	1	160	Mean Difference (IV, Fixed, 95% CI)	0.42 [0.11, 0.73]
15 Infant injuries	1	160	Risk Ratio (M-H, Fixed, 95% CI)	0.54 [0.31, 0.92]
15.1 Infant injuries at 9 to 16 weeks	1	160	Risk Ratio (M-H, Fixed, 95% CI)	0.54 [0.31, 0.92]
16 Marital discord at 4 and 24 weeks postpartum	1		Mean Difference (IV, Fixed, 95% CI)	Subtotals only
16.1 Marital discord at 0 to 8 weeks	1	31	Mean Difference (IV, Fixed, 95% CI)	-3.20 [-16.93, 10.53]
16.2 Marital discord at 17 to 24 weeks	1	29	Mean Difference (IV, Fixed, 95% CI)	-7.90 [-21.52, 5.72]

Comparison 2. Psychosocial interventions versus usual care - variations in intervention type

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Antenatal and postnatal classes	2	311	Risk Ratio (M-H, Fixed, 95% CI)	1.02 [0.61, 1.72]
1.1 Depressive symptomatology at final study assessment	2	311	Risk Ratio (M-H, Fixed, 95% CI)	1.02 [0.61, 1.72]
2 Postpartum professional-based home visits	2	1663	Risk Ratio (M-H, Fixed, 95% CI)	0.68 [0.55, 0.84]
2.1 Depressive symptomatology at final study assessment	2	1663	Risk Ratio (M-H, Fixed, 95% CI)	0.68 [0.55, 0.84]
3 Postpartum lay-based home visits	1	481	Risk Ratio (M-H, Fixed, 95% CI)	0.89 [0.62, 1.27]
3.1 Depressive symptomatology at final study assessment	1	481	Risk Ratio (M-H, Fixed, 95% CI)	0.89 [0.62, 1.27]
4 Early postpartum follow up	1	475	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.56, 1.48]

4.1 Depressive symptomatology at final study assessment	1	475	Risk Ratio (M-H, Fixed, 95% CI)	0.91 [0.56, 1.48]
	1	025		1 2 / [0 07 1 05]
5 Continuity of care	1	935	Risk Ratio (M-H, Fixed, 95% CI)	1.34 [0.97, 1.85]
5.1 Depressive	1	935	Risk Ratio (M-H, Fixed, 95% CI)	1.34 [0.97, 1.85]
symptomatology at final study				
assessment				

Comparison 3. Psychological Interventions versus usual care - variations in intervention type

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Psychological debriefing	5		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
1.1 Depressive symptomatology at final study assessment - all trials	5	3051	Risk Ratio (M-H, Random, 95% CI)	0.57 [0.31, 1.04]
1.2 Depressive symptomatology at final study assessment - sensitivity analysis	4	2535	Risk Ratio (M-H, Random, 95% CI)	0.51 [0.23, 1.14]
2 Interpersonal psychotherapy	2	72	Risk Ratio (M-H, Random, 95% CI)	0.31 [0.04, 2.52]
2.1 Depressive symptomatology at final study assessment	2	72	Risk Ratio (M-H, Random, 95% CI)	0.31 [0.04, 2.52]

Comparison 4. All interventions versus usual care - variations in intervention mode

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Individually based interventions	11		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
1.1 Depressive symptomatology at 0 to 8 weeks - all trials	7	3963	Risk Ratio (M-H, Random, 95% CI)	0.64 [0.40, 1.01]
1.2 Depressive symptomatology at 0 to 8 weeks - sensitivity analysis	6	3447	Risk Ratio (M-H, Random, 95% CI)	0.60 [0.34, 1.05]
1.3 Depressive symptomatology at 9 to 16 weeks - all trials	4	2241	Risk Ratio (M-H, Random, 95% CI)	0.71 [0.45, 1.12]
1.4 Depressive symptomatology at 9 to 16 weeks - sensitivity analysis	3	738	Risk Ratio (M-H, Random, 95% CI)	0.67 [0.31, 1.48]
1.5 Depressive symptomatology at 17 to 24 weeks	5	3484	Risk Ratio (M-H, Random, 95% CI)	0.98 [0.82, 1.17]

1.6 Depressive symptomatology at final study assessment - all trials	11	6642	Risk Ratio (M-H, Random, 95% CI)	0.76 [0.59, 1.00]
1.7 Depressive symptomatology at final study assessment - sensitivity analysis	7	3667	Risk Ratio (M-H, Random, 95% CI)	0.68 [0.43, 1.09]
2 Group-based interventions	4		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
2.1 Depressive symptomatology at 0 to 8 weeks	1	128	Risk Ratio (M-H, Random, 95% CI)	0.73 [0.31, 1.69]
2.2 Depressive symptomatology at 9 to 16 weeks - all trials	4	1085	Risk Ratio (M-H, Random, 95% CI)	0.93 [0.54, 1.59]
2.3 Depressive symptomatology at 9 to 16 weeks - sensitivity analysis	3	1050	Risk Ratio (M-H, Random, 95% CI)	1.08 [0.74, 1.57]
2.4 Depressive symptomatology at 17 to 24 weeks	2	830	Risk Ratio (M-H, Random, 95% CI)	1.20 [0.85, 1.71]
2.5 Depressive symptomatology at final study assessment	4	1055	Risk Ratio (M-H, Random, 95% CI)	1.03 [0.65, 1.63]

Comparison 5. All interventions versus usual care - variations in intervention onset

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Interventions with antenatal and postnatal components	4	1283	Risk Ratio (M-H, Fixed, 95% CI)	1.21 [0.93, 1.59]
1.1 Depressive symptomatology at final study assessment	4	1283	Risk Ratio (M-H, Fixed, 95% CI)	1.21 [0.93, 1.59]
2 Interventions with postnatal only component	10		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
2.1 Depressive symptomatology at final study assessment - all trials	10	6379	Risk Ratio (M-H, Random, 95% CI)	0.76 [0.58, 0.98]
2.2 Depressive symptomatology at final study assessment - sensitivity analysis	5	2695	Risk Ratio (M-H, Random, 95% CI)	0.56 [0.29, 1.06]

Outcome or subgroup title	No. of No. of studies participants		Statistical method	Effect size	
1 Single-contact interventions	4		Risk Ratio (M-H, Random, 95% CI)	Subtotals only	
1.1 Depressive symptomatology at 0 to 8 weeks	2	1756	Risk Ratio (M-H, Random, 95% CI)	0.39 [0.07, 2.16]	
1.2 Depressive symptomatology at 9 to 16 weeks	1	475	Risk Ratio (M-H, Random, 95% CI)	1.21 [0.78, 1.85]	
1.3 Depressive symptomatology at 17 to 24 weeks	3	2966	Risk Ratio (M-H, Random, 95% CI)	1.02 [0.82, 1.26]	
1.4 Depressive symptomatology at final study assessment - all trials	4	2898	Risk Ratio (M-H, Random, 95% CI)	0.70 [0.38, 1.27]	
1.5 Depressive symptomatology at final study assessment - sensitivity analysis	3	2432	Risk Ratio (M-H, Random, 95% CI)	0.61 [0.26, 1.45]	
2 Multiple-contact interventions	11		Risk Ratio (M-H, Random, 95% CI)	Subtotals only	
2.1 Depressive symptomatology at 0 to 8 weeks - all trials	6	2274	Risk Ratio (M-H, Random, 95% CI)	0.77 [0.50, 1.17]	
2.2 Depressive symptomatology at 0 to 8 weeks - sensitivity analysis	5	1758	Risk Ratio (M-H, Random, 95% CI)	0.75 [0.44, 1.27]	
2.3 Depressive symptomatology at 9 to 16 weeks - all trials	7	2828	Risk Ratio (M-H, Random, 95% CI)	0.72 [0.52, 1.01]	
2.4 Depressive symptomatology at 9 to 16 weeks - sensitivity analysis	5	1290	Risk Ratio (M-H, Random, 95% CI)	0.76 [0.49, 1.17]	
2.5 Depressive symptomatology at 17 to 24 weeks	4	1348	Risk Ratio (M-H, Random, 95% CI)	1.02 [0.80, 1.31]	
2.6 Depressive symptomatology at final study assessment - all trials	11	4790	Risk Ratio (M-H, Random, 95% CI)	0.84 [0.66, 1.08]	
2.7 Depressive symptomatology at final study assessment - sensitivity analysis	6	1546	Risk Ratio (M-H, Random, 95% CI)	0.84 [0.53, 1.34]	

Comparison 6. All interventions versus usual care - variations in intervention duration

Comparison 7. All interventions versus usual care - variations in risk status

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Interventions for at-risk women	7	1162	Risk Ratio (M-H, Fixed, 95% CI)	0.67 [0.51, 0.89]
1.1 Depressive symptomatology at final study assessment	7	1162	Risk Ratio (M-H, Fixed, 95% CI)	0.67 [0.51, 0.89]
2 Interventions for general population	8		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
2.1 Depressive symptomatology at final study assessment - all trials	8	6535	Risk Ratio (M-H, Random, 95% CI)	0.87 [0.66, 1.16]
2.2 Depressive symptomatology at final study assessment - sensitivity analysis	4	3367	Risk Ratio (M-H, Random, 95% CI)	0.80 [0.45, 1.40]

Analysis I.I. Comparison I All interventions versus usual care - various study outcomes, Outcome I Depressive symptomatology at final study assessment (variously defined).

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: I Depressive symptomatology at final study assessment (variously defined)

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Random,95% CI		M-H,Random,95% Cl
I Depressive symptomatolog	y - all trials				
Armstrong 1999	13/80	18/80	-	6.3 %	0.72 [0.38, 1.37]
Brugha 2000	15/94	18/96	+	6.5 %	0.85 [0.46, 1.59]
Gunn 1998	27/232	31/243	+	7.8 %	0.91 [0.56, 1.48]
Lavender 1998	5/58	31/56		4.6 %	0.16 [0.07, 0.37]
MacArthur 2002	5/80	149/702		10.3 %	0.68 [0.54, 0.84]
Morrell 2000	48/252	49/229	-	9.1 %	0.89 [0.62, 1.27]
Reid 2002	49/339	46/370	+	8.9 %	1.16 [0.80, 1.69]
Small 2000	81/467	65/450	+	9.6 %	1.20 [0.89, 1.62]
Stamp 1995	9/60	6/61		4.1 %	I.53 [0.58, 4.02]
Waldenstrom 2000	74/464	56/471	÷	9.4 %	1.34 [0.97, 1.85]

0.001 0.01 0.1 1 10 100 1000

Favours treatment Favours control

(Continued ...)

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	(Continued) Risk Ratio M-H,Random,95% Cl
Gamble 2003	4/50	17/53		3.8 %	0.25 [0.09, 0.69]
Gorman 2002	3/20	4/17		2.5 %	0.64 [0.17, 2.46]
Priest 2003	37/696	42/705	+	8.4 %	0.89 [0.58, 1.37]
Tam 2003	26/261	35/255	-	7.9 %	0.73 [0.45, 1.17]
Zlotnick 2001	0/17	6/18	•	0.7 %	0.08 [0.00, 1.34]
Subtotal (95% CI)	3891	3806	•	100.0 %	0.81 [0.65, 1.02]
Armstrong 1999 Brugha 2000 Gamble 2003 Gorman 2002	13/80 15/94 4/50 3/20	18/80 18/96 17/53 4/17		11.4 % 11.8 % 6.9 % 4.6 %	0.72 [0.38, 1.37] 0.85 [0.46, 1.59] 0.25 [0.09, 0.69] 0.64 [0.17, 2.46]
Lavender 1998	5/58	31/56	-	8.4 %	0.16 [0.07, 0.37]
Priest 2003	37/696	42/705	+	15.1 %	0.89 [0.58, 1.37]
Small 2000	81/467	65/450	-	17.4 %	1.20 [0.89, 1.62]
Stamp 1995	9/60	6/61		7.3 %	1.53 [0.58, 4.02]
Waldenstrom 2000	74/464	56/471	-	17.0 %	1.34 [0.97, 1.85]
Subtotal (95% CI) Total events: 241 (Treatment), Heterogeneity: Tau ² = 0.23; C Test for overall effect: Z = 1.37	hi ² = 32.33, df = 8 (P	1989 = 0.00008); I ² =75%	•	100.0 %	0.76 [0.52, 1.12]

0.001 0.01 0.1 1 10 100 1000

Favours treatment Favours control

Analysis 1.2. Comparison I All interventions versus usual care - various study outcomes, Outcome 2 Depressive symptomatology at final study assessment (EPDS > 12).

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 2 Depressive symptomatology at final study assessment (EPDS > 12)

	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratio M-H,Random,95% Cl
I Depressive symptomatology	- all trials				
Armstrong 1999	13/80	18/80		7.6 %	0.72 [0.38, 1.37]
Gamble 2003	4/50	15/53		3.8 %	0.28 [0.10, 0.79]
Gorman 2002	3/15	3/15		2.2 %	1.00 [0.24, 4.18]
Gunn 1998	27/232	31/243	-	10.5 %	0.91 [0.56, 1.48]
MacArthur 2002	5/80	149/702	+	17.1 %	0.68 [0.54, 0.84]
Morrell 2000	48/252	49/229		13.5 %	0.89 [0.62, 1.27]
Priest 2003	37/696	42/705		11.7 %	0.89 [0.58, 1.37]
Small 2000	81/467	65/450		15.0 %	1.20 [0.89, 1.62]
Stamp 1995	9/60	6/61		4.2 %	1.53 [0.58, 4.02]
Waldenstrom 2000	74/464	56/471		14.4 %	1.34 [0.97, 1.85]
Subtotal (95% CI)	3117	3009	+	100.0 %	0.91 [0.73, 1.15]
, ,	. ,	= 0.01); I ² =60%			
Total events: 411 (Treatment), Heterogeneity: Tau ² = 0.07; Cf Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999	$hi^2 = 22.28, df = 9 (P$			12.9 %	0.72 [0.38, 1.37]
Heterogeneity: $Tau^2 = 0.07$; Ch Test for overall effect: Z = 0.79	ni ² = 22.28, df = 9 (P (P = 0.43)	= 0.01); l ² =60% 18/80 15/53		12.9 %	0.72 [0.38, 1.37] 0.28 [0.10, 0.79]
Heterogeneity: Tau ² = 0.07; Cł Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999 Gamble 2003	hi ² = 22.28, df = 9 (P (P = 0.43) I 3/80	18/80			0.28 [0.10, 0.79]
Heterogeneity: Tau ² = 0.07; Cr Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999	h ² = 22.28, df = 9 (P (P = 0.43) 13/80 4/50	18/80		6.5 %	
Heterogeneity: Tau ² = 0.07; Cr Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999 Gamble 2003 Gorman 2002	h ² = 22.28, df = 9 (P (P = 0.43) 13/80 4/50 3/15	18/80 15/53 3/15		6.5 % 3.7 %	0.28 [0.10, 0.79] 1.00 [0.24, 4.18] 0.89 [0.58, 1.37]
Heterogeneity: Tau ² = 0.07; Cł Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999 Gamble 2003 Gorman 2002 Priest 2003	h ² = 22.28, df = 9 (P (P = 0.43) 13/80 4/50 3/15 37/696	18/80 15/53 3/15 42/705		6.5 % 3.7 % 19.8 %	0.28 [0.10, 0.79] 1.00 [0.24, 4.18] 0.89 [0.58, 1.37] 1.20 [0.89, 1.62]
Heterogeneity: Tau ² = 0.07; Cr Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999 Gamble 2003 Gorman 2002 Priest 2003 Small 2000	h ² = 22.28, df = 9 (P (P = 0.43) 13/80 4/50 3/15 37/696 81/467	18/80 15/53 3/15 42/705 65/450		6.5 % 3.7 % 19.8 % 25.4 %	0.28 [0.10, 0.79] 1.00 [0.24, 4.18] 0.89 [0.58, 1.37] 1.20 [0.89, 1.62] 1.53 [0.58, 4.02]
Heterogeneity: Tau ² = 0.07; Cr Test for overall effect: Z = 0.79 2 Sensitivity analysis Armstrong 1999 Gamble 2003 Gorman 2002 Priest 2003 Small 2000 Stamp 1995	hi ² = 22.28, df = 9 (P (P = 0.43) 13/80 4/50 3/15 37/696 81/467 9/60 74/464 1832 205 (Control)	18/80 15/53 3/15 42/705 65/450 6/61 56/471 1835		6.5 % 3.7 % 19.8 % 25.4 % 7.2 %	0.28 [0.10, 0.79]

Favours treatment Favours control

Analysis I.3. Comparison I All interventions versus usual care - various study outcomes, Outcome 3 Mean depression scores at final study assessment.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 3 Mean depression scores at final study assessment

Mean(SD) 5.75 (5.51) 7.9 (5.2) 5.87 (5.37) 6.4 (0.98)	N 80 15 243	Mean(SD) 6.64 (5.58) 8 (5.6)	IV,Random,95% CI	9.8 %	IV,Random,95% C -0.89 [-2.61, 0.83
7.9 (5.2) 5.87 (5.37)	15	× ,		9.8 %	-0.89 [-2.61, 0.83
7.9 (5.2) 5.87 (5.37)	15	× ,		9.8 %	-0.89 [-2.61, 0.83
5.87 (5.37)		8 (5.6)			
. ,	243			3.8 %	-0.10 [-3.97, 3.77
6.4 (0.98)		6.08 (5.14)		13.4 %	-0.21 [-1.16, 0.74
```	743	8.06 (1.41)	-	15.9 %	-1.66 [ -1.78, -1.54
6.6 (5.1)	233	6.7 (5.6)		13.4 %	-0.10 [ -1.05, 0.85
5.3 (5.4)	370	5.3 (4.84)	-	14.2 %	0.0 [ -0.76, 0.76
7.16 (5.68)	450	6.72 (5.5)	+	14.4 %	0.44 [ -0.28, 1.16
3.3 (2.9)	255	3.5 (3)	-	15.1 %	-0.20 [ -0.71, 0.31
	2389		•	100.0 %	-0.36 [ -1.21, 0.48
0, df = 7 (P<0.000	01); I ² =92%	6			
))					
5.75 (5.51)	80	6.64 (5.58)		35.0 %	-0.89 [ -2.61, 0.83
7.9 (5.2)	15	8 (5.6)		13.7 %	-0.10 [ -3.97, 3.77
7.16 (5.68)	450	6.72 (5.5)	-	51.3 %	0.44 [ -0.28, 1.16
	545		•	100.0 %	0.23 [ -0.43, 0.89
df = 2 (P = 0.37); I	2 =0.0%				
')					
	5.3 (5.4) 7.16 (5.68) 3.3 (2.9) 0, df = 7 (P<0.000 0) 5.75 (5.51) 7.9 (5.2) 7.16 (5.68)	5.3 (5.4) 370 7.16 (5.68) 450 3.3 (2.9) 255 <b>2389</b> 0, df = 7 (P<0.00001); $l^2 = 929$ 0) 5.75 (5.51) 80 7.9 (5.2) 15 7.16 (5.68) 450 <b>545</b> df = 2 (P = 0.37); $l^2 = 0.0\%$	5.3 (5.4)   370   5.3 (4.84)  7.16 (5.68)   450   6.72 (5.5)  3.3 (2.9)   255   3.5 (3)  2389  0, df = 7 (P<0.00001); l2 = 92%  0)  5.75 (5.51)   80   6.64 (5.58)  7.9 (5.2)   15   8 (5.6)  7.16 (5.68)   450   6.72 (5.5)  545  df = 2 (P = 0.37); l2 = 0.0%	5.3 (5.4) 370 5.3 (4.84) 7.16 (5.68) 450 6.72 (5.5) 3.3 (2.9) 255 3.5 (3) <b>2389</b> 0, df = 7 (P<0.00001); $ ^2 = 92\%$ 0) 5.75 (5.51) 80 6.64 (5.58) 7.9 (5.2) 15 8 (5.6) 7.16 (5.68) 450 6.72 (5.5) <b>545</b> df = 2 (P = 0.37); $ ^2 = 0.0\%$	$5.3 (5.4) 370 5.3 (4.84) + 14.2 \%$ $7.16 (5.68) 450 6.72 (5.5) + 14.4 \%$ $3.3 (2.9) 255 3.5 (3) + 15.1 \%$ $2389 + 100.0 \%$ $0, df = 7 (P<0.00001); 1^2 = 92\%$ $0)$ $5.75 (5.51) 80 6.64 (5.58) + 35.0 \%$ $7.9 (5.2) 15 8 (5.6) + 13.7 \%$ $7.16 (5.68) 450 6.72 (5.5) + 51.3 \%$ $6df = 2 (P = 0.37); 1^2 = 0.0\%$

Favours treatment Favours control

#### Analysis I.4. Comparison I All interventions versus usual care - various study outcomes, Outcome 4 Depressive symptomatology at 8, 16, 24 weeks postpartum (variously defined).

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 4 Depressive symptomatology at 8, 16, 24 weeks postpartum (variously defined)

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratio M-H,Random,95% Cl
I Depressive symptomatology	/ at 0 to 8 weeks - all t	trials			
Armstrong 1999	5/86	18/88		8.0 %	0.28 [ 0.11, 0.73 ]
Gorman 2002	0/20	5/20	<u>+</u>	1.3 %	0.09 [ 0.01, 1.54 ]
Lavender 1998	5/58	31/56	+	8.9 %	0.16 [ 0.07, 0.37 ]
Morrell 2000	49/276	48/266	+	18.6 %	0.98 [ 0.69, 1.41 ]
Priest 2003	54/809	63/833	-	18.8 %	0.88 [ 0.62, 1.25 ]
Stamp 1995	8/64	/64		9.2 %	0.73 [ 0.31, 1.69 ]
Tam 2003	26/261	35/255	-	15.9 %	0.73 [ 0.45, 1.17 ]
Waldenstrom 2000	74/464	56/471	-	19.4 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI)	2038	2053	•	100.0 %	0.65 [ 0.43, 1.00 ]
2 Depressive symptomatology Armstrong 1999	5/86	18/88		9.5 %	0.28 [ 0.11, 0.73 ]
Heterogeneity: $Tau^2 = 0.24$ ; C Test for overall effect: $Z = 1.9$		$= 0.00005);  ^2 = 78\%$			
5					
Gorman 2002	0/20	5/20		1.5 %	0.09 [ 0.01, 1.54 ]
Lavender 1998	5/58	31/56	-	10.5 %	0.16 [ 0.07, 0.37 ]
Morrell 2000	49/276	48/266	+	22.1 %	0.98 [ 0.69, 1.41 ]
Priest 2003	54/809	63/833	-	22.3 %	0.88 [ 0.62, 1.25 ]
Stamp 1995	8/64	11/64	-	11.0 %	0.73 [ 0.31, 1.69 ]
Waldenstrom 2000	74/464	56/471	-	23.1 %	1.34 [ 0.97, 1.85 ]
<b>Subtotal (95% CI)</b> Total events: 195 (Treatment), Heterogeneity: Tau ² = 0.30; C Test for overall effect: Z = 1.8	$Chi^2 = 30.78, df = 6 (P$	<b>1798</b> = 0.00003);   ² = 8   %	•	100.0 %	0.62 [ 0.38, 1.03 ]
3 Depressive symptomatology	vat 9 to 16 weeks - all	trials			
Armstrong 1999	13/80	18/80	-	12.5 %	0.72 [ 0.38, 1.37 ]
Brugha 2000	15/94	18/96	+	12.9 %	0.85 [ 0.46, 1.59 ]
			0.001 0.01 0.1 10 100 1000 Favours treatment Favours control		

(Continued . . . )

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	( Continued Risk Ratio M-H,Random,95% Cl
Gamble 2003	4/50	17/53		7.3 %	0.25 [ 0.09, 0.69 ]
Gunn 1998	38/232	33/243	-	17.1 %	.2  [ 0.78,  .85 ]
MacArthur 2002	115/801	149/702	-	21.8 %	0.68 [ 0.54, 0.84 ]
Reid 2002	55/344	46/388	_	18.6 %	1.35 [ 0.94, 1.94 ]
			_		
Stamp 1995	7/63	10/65		8.5 %	0.72 [ 0.29, 1.78 ]
Zlotnick 2001	0/17	6/18		1.3 %	0.08 [ 0.00, 1.34 ]
Subtotal (95% CI) Total events: 247 (Treatment), Heterogeneity: Tau ² = 0.13; C Test for overall effect: Z = 1.30	hi ² = 21.34, df = 7 (P	<b>1645</b> = 0.003); I ² =67%	•	100.0 %	0.80 [ 0.56, 1.12 ]
4 Depressive symptomatology					
Armstrong 1999	13/80	18/80	1	16.3 %	0.72 [ 0.38, 1.37 ]
Brugha 2000	15/94	18/96	-	16.8 %	0.85 [ 0.46, 1.59 ]
Gamble 2003	4/50	17/53		9.4 %	0.25 [ 0.09, 0.69 ]
Gunn 1998	38/232	33/243	+	22.2 %	1.21 [ 0.78, 1.85 ]
Reid 2002	55/344	46/388	-	24.2 %	1.35 [ 0.94, 1.94 ]
Stamp 1995	7/63	10/65		11.1 %	0.72 [ 0.29, 1.78 ]
Subtotal (95% CI)	863	925	+	100.0 %	0.87 [ 0.59, 1.28 ]
Total events: 132 (Treatment), Heterogeneity: Tau ² = 0.12; C Test for overall effect: Z = 0.71 5 Depressive symptomatology	hi ² = 12.24, df = 5 (P I (P = 0.48) at 17 to 24 weeks				
Gorman 2002	3/20	4/17		4.5 %	0.64 [ 0.17, 2.46 ]
Gunn 1998	27/232	31/243	-	15.1 %	0.91 [ 0.56, 1.48 ]
Morrell 2000	48/252	49/229	-	18.0 %	0.89 [ 0.62, 1.27 ]
Priest 2003	55/777	65/797	+	18.2 %	0.87 [ 0.61, 1.23 ]
Reid 2002	49/339	46/370	+	17.5 %	1.16 [ 0.80, 1.69 ]
Small 2000	81/467	65/450	-	19.2 %	1.20 [ 0.89, 1.62 ]
Stamp 1995	9/60	6/61		7.4 %	1.53 [ 0.58, 4.02 ]
Subtotal (95% CI)	2147	2167	•	100.0 %	1.02 [ 0.87, 1.19 ]
Total events: 272 (Treatment), Heterogeneity: Tau ² = 0.0; Chi Test for overall effect: $Z = 0.25$	$i^2 = 4.34$ , df = 6 (P =	0.63); I ² =0.0%			

0.001 0.01 0.1 10 100 1000 Favours treatment Favours control

#### Analysis 1.5. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 5 Depressive symptomatology at 8, 16, 24 weeks postpartum (EPDS > 12).

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 5 Depressive symptomatology at 8, 16, 24 weeks postpartum (EPDS > 12)

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratio M-H,Random,95% Cl
I Depressive symptomatology					
Armstrong 1999	5/86	18/88		8.1 %	0.28 [ 0.11, 0.73 ]
Gorman 2002	3/18	2/13		3.1 %	1.08 [ 0.21, 5.59 ]
Morrell 2000	49/276	48/266	+	25.5 %	0.98 [ 0.69, 1.41 ]
Priest 2003	54/809	63/833	-	26.1 %	0.88 [ 0.62, 1.25 ]
Stamp 1995	8/64	11/64		9.6 %	0.73 [ 0.31, 1.69 ]
Waldenstrom 2000	74/464	56/471	-	27.6 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI)	1717	1735	•	100.0 %	0.90 [ 0.65, 1.25 ]
Total events: 193 (Treatment), Heterogeneity: Tau ² = 0.08; Cł Test for overall effect: Z = 0.61 2 Depressive symptomatology	$hi^2 = 11.08, df = 5 (P)$ (P = 0.54)	<i>,.</i>			
Armstrong 1999	13/80	18/80		6.6 %	0.72 [ 0.38, 1.37 ]
Gamble 2003	4/50	17/53		8.4 %	0.25 [ 0.09, 0.69 ]
Gunn 1998	38/232	33/243	-	25.9 %	1.21 [ 0.78, 1.85 ]
MacArthur 2002	5/80	149/702	-	38.9 %	0.68 [ 0.54, 0.84 ]
Stamp 1995	7/63	10/65		10.2 %	0.72 [ 0.29, 1.78 ]
<b>Subtotal (95% CI)</b> Total events: 177 (Treatment), Heterogeneity: Tau ² = 0.10; Cf Test for overall effect: Z = 1.67	$hi^2 = 9.97, df = 4 (P = 1)$	<b>1143</b> = 0.04);   ² =60%	•	100.0 %	0.72 [ 0.49, 1.06 ]
3 Depressive symptomatology		, ,			
Armstrong 1999	13/80	18/80		27.2 %	0.72 [ 0.38, 1.37 ]
Gamble 2003	4/50	17/53		13.7 %	0.25 [ 0.09, 0.69 ]
Gunn 1998	38/232	33/243	-	42.4 %	1.21 [ 0.78, 1.85 ]
Stamp 1995	7/63	10/65		16.7 %	0.72 [ 0.29, 1.78 ]
<b>Subtotal (95% CI)</b> Total events: 62 (Treatment), 7 Heterogeneity: Tau ² = 0.23; Ch	, ,	<b>441</b> = 0.04); I ² =65%	•	100.0 %	0.70 [ 0.39, 1.28 ]

(Continued . . . )

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	( Continued) Risk Ratio M-H,Random,95% Cl
Test for overall effect: $Z = I$ .	I6 (P = 0.25)				
4 Depressive symptomatolog	gy at 17 to 24 weeks				
Gorman 2002	3/15	3/15		3.5 %	1.00 [ 0.24, 4.18 ]
Gunn 1998	27/232	31/243	+	17.5 %	0.91 [ 0.56, 1.48 ]
Morrell 2000	48/252	49/229	+	23.0 %	0.89 [ 0.62, 1.27 ]
Priest 2003	55/777	65/797	+	23.4 %	0.87 [ 0.61, 1.23 ]
Small 2000	81/467	65/450	-	25.7 %	1.20 [ 0.89, 1.62 ]
Stamp 1995	9/60	6/61		6.9 %	I.53 [ 0.58, 4.02 ]
Subtotal (95% CI)	1803	1795	•	100.0 %	1.00 [ 0.84, 1.19 ]
Total events: 223 (Treatment)	), 219 (Control)				
Heterogeneity: $Tau^2 = 0.0$ ; C	hi ² = 3.36, df = 5 (P =	0.65); l ² =0.0%			
Test for overall effect: $Z = 0.0$	00 (P = 1.0)				
			0.01 0.1 1 10 100		
			Favours treatment Favours control		

#### Analysis I.6. Comparison I All interventions versus usual care - various study outcomes, Outcome 6 Maternal health service contact at final study assessment.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 6 Maternal health service contact at final study assessment

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Maternal health service	contact				
Brugha 2000	13/94	11/96		100.0 %	1.21 [ 0.57, 2.56 ]
Total (95% CI)	94	96	-	100.0 %	1.21 [ 0.57, 2.56 ]
Total events: 13 (Treatmer	nt), II (Control)				
Heterogeneity: not applica	able				
Test for overall effect: Z =	0.49 (P = 0.62)				
			0.1 0.2 0.5 1 2 5 10		
			Favours treatment Favours control		

Psychosocial and psychological interventions for preventing postpartum depression (Review) Copyright © 2008 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### Analysis 1.7. Comparison I All interventions versus usual care - various study outcomes, Outcome 7 Maternal-infant attachment at 8, 16, 24 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 7 Maternal-infant attachment at 8, 16, 24 weeks postpartum

Study or subgroup	Treatment N	Mean(SD)	Control N	Mean(SD)	Mean Difference IV,Fixed,95% CI	Weight	Mean Difference IV,Fixed,95% CI
I Maternal-infant attachme	ent at 0 to 8 we	eks					
Armstrong 1999	86	12.33 (3.69)	88	12.43 (3.61)	-	100.0 %	-0.10 [ -1.18, 0.98 ]
Subtotal (95% CI)	86		88		+	100.0 %	-0.10 [ -1.18, 0.98 ]
Heterogeneity: not applica	ble						
Test for overall effect: Z =	0.18 (P = 0.86)	)					
2 Maternal-infant attachme	ent at 9 to 16 w	veeks					
Armstrong 1999	80	12.05 (3.47)	80	12.78 (3.67)		100.0 %	-0.73 [ -1.84, 0.38 ]
Subtotal (95% CI)	80		80		•	100.0 %	-0.73 [ -1.84, 0.38 ]
Heterogeneity: not applica	ble						
Test for overall effect: Z =	I.29 (P = 0.20)	)					
Test for subgroup difference	tes: $Chi^2 = 0.63$	, df = 1 (P = 0.4	3), l ² =0.0%				
				- I C	-5 0 5	10	

-10 -5 0 5

Favours treatment Favours control

#### Analysis 1.8. Comparison I All interventions versus usual care - various study outcomes, Outcome 8 Maternal attitudes toward motherhood at 8, 16, 24 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 8 Maternal attitudes toward motherhood at 8, 16, 24 weeks postpartum

Study or subgroup	Treatment		Control		Mean Difference	Weight	Mean Difference
	Ν	Mean(SD)	Ν	Mean(SD)	IV,Fixed,95% CI		IV,Fixed,95% CI
I Restriction of role at 0 to	o 8 weeks						
Armstrong 1999	86	19.47 (5.88)	88	19.6 (5.88)		100.0 %	-0.13 [ -1.88, 1.62 ]
Subtotal (95% CI)	86		88		+	100.0 %	-0.13 [ -1.88, 1.62 ]
Heterogeneity: not applicat	ole						
Test for overall effect: $Z =$	0.15 (P = 0.88)	)					
2 Restriction of role at 9 to	16 weeks						
Armstrong 1999	80	18.28 (6)	80	19.4 (5.84)	-	100.0 %	-1.12 [ -2.95, 0.71 ]
Subtotal (95% CI)	80		80		•	100.0 %	-1.12 [ -2.95, 0.71 ]
Heterogeneity: not applicat	ole						
Test for overall effect: Z =	1.20 (P = 0.23)	)					
Test for subgroup difference	es: Chi ² = 0.59	P, df = 1 (P = 0.44)	), l ² =0.0%				
						1	
				-10	-5 0 5	10	
				Favours	treatment Favours co	ntrol	

#### Analysis I.9. Comparison I All interventions versus usual care - various study outcomes, Outcome 9 Maternal anxiety at 8, 16, 24 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 9 Maternal anxiety at 8, 16, 24 weeks postpartum

Study or subgroup	Treatment n/N	Control n/N			Risk Ratio ×ed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Maternal anxiety at 0 to 8 v	veeks						
Lavender 1998	4/58	28/56				100.0 %	0.14 [ 0.05, 0.37 ]
Subtotal (95% CI)	58	56		٠		100.0 %	0.14 [ 0.05, 0.37 ]
Total events: 4 (Treatment), 2	8 (Control)						
Heterogeneity: not applicable							
Test for overall effect: $Z = 3.9$	96 (P = 0.000076)						
						1	
			0.01	0.1	I IO I	100	
			Favours t	reatment	Favours cor	ntrol	

#### Analysis 1.10. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 10 Competence in mothering at 8 and 16 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 10 Competence in mothering at 8 and 16 weeks postpartum

Study or subgroup	Treatment N	Mean(SD)	Control N	Mean(SD)	Mean Differ IV,Fixed,95%		Mean Difference IV,Fixed,95% Cl
I Maternal competence at	t 0 to 8 weeks						
Armstrong 1999	86	28.37 (7.1)	88	28.03 (6.46)	-	100.0 %	0.34 [ -1.68, 2.36 ]
Subtotal (95% CI)	86		88		-	100.0 %	0.34 [ -1.68, 2.36 ]
Heterogeneity: not applica	ble						
Test for overall effect: Z =	0.33 (P = 0.74)						
2 Maternal competence at	9 to 16 weeks						
Armstrong 1999	80	27.74 (6.9)	80	29.2 (8.4)		100.0 %	-1.46 [ -3.84, 0.92 ]
Subtotal (95% CI)	80		80		-	100.0 %	-1.46 [ -3.84, 0.92 ]
Heterogeneity: not applica	ble						
Test for overall effect: Z =	1.20 (P = 0.23)						
Test for subgroup difference	ces: Chi ² = 1.28,	df = 1 (P = 0.2	6), I ² =22%				
				-10	-5 0	5 10	
				Favours	treatment Favo	ours control	

#### Analysis 1.11. Comparison I All interventions versus usual care - various study outcomes, Outcome I I General physical and mental health (SF-36) at final study assessment.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: II General physical and mental health (SF-36) at final study assessment

Study or subgroup	Treatment N	Mean(SD)	Control N	Mean(SD)	Mean Difference IV,Fixed,95% CI	Weight	Mean Difference IV,Fixed,95% Cl
I Physical functioning							
Gunn 1998	232	86.8 (17.8)	243	86.5 (16.3)	-	15.2 %	0.30 [ -2.77, 3.37 ]
Morrell 2000	258	89.8 (16.8)	230	91.2 (15.1)	-	18.0 %	-1.40 [ -4.23, 1.43 ]
Reid 2002	339	93.7 (11.69)	370	92.7 (14.04)	•	40.1 %	1.00 [ -0.90, 2.90 ]
Small 2000	467	86.1 (17.4)	450	85.73 (18.44)	+	26.7 %	0.37 [ -1.95, 2.69 ]
Subtotal (95% CI)	1296		1293			100.0 %	0.29 [ -0.91, 1.49 ]
Heterogeneity: Chi ² = 1.91	, df = 3 (P =	0.59); l ² =0.0%					
Test for overall effect: $Z =$	0.48 (P = 0.63	3)					
2 Role functioning (physical	l)						
Gunn 1998	232	73.7 (35.7)	242	72.1 (36.1)	+	14.1 %	1.60 [ -4.86, 8.06 ]
Morrell 2000	259	80.2 (32.5)	229	82.1 (32.6)	+	17.6 %	-1.90 [ -7.69, 3.89 ]
Reid 2002	339	87.6 (26)	370	87.9 (26.21)	•	39.9 %	-0.30 [ -4.15, 3.55 ]
Small 2000	467	73.86 (35.1)	450	76.24 (35.29)	-	28.4 %	-2.38 [ -6.94, 2.18 ]
Subtotal (95% CI)	1297		1291		•	100.0 %	-0.90 [ -3.33, 1.52 ]
Heterogeneity: Chi ² = 1.19	9, df = 3 (P =	0.76); l ² =0.0%					
Test for overall effect: $Z =$	0.73 (P = 0.47	7)					
3 Bodily pain							
Gunn 1998	232	77.8 (22.9)	243	75.9 (23.1)	-	16.2 %	1.90 [ -2.24, 6.04 ]
Morrell 2000	256	81 (22.7)	232	82.8 (23.2)	-	16.7 %	-1.80 [ -5.88, 2.28 ]
Reid 2002	339	87.3 (17.9)	370	85.9 (19.32)	+	36.9 %	1.40 [ -1.34, 4.14 ]
Small 2000	467	77.7 (23.22)	450	78.6 (23.55)	-	30.2 %	-0.90 [ -3.93, 2.13 ]
Subtotal (95% CI)	1294		1295			100.0 %	0.25 [ -1.41, 1.92 ]
Heterogeneity: $Chi^2 = 2.81$	, df = 3 (P =	0.42); l ² =0.0%					
Test for overall effect: Z =	0.30 (P = 0.77	7)					
4 Mental health							
Gunn 1998	232	70.3 (19.7)	243	72.  ( 8. )	-	16.0 %	-1.80[-5.21, 1.61]
Morrell 2000	254	72.8 (17.3)	227	74 (17.5)	-	19.2 %	-1.20 [ -4.32, 1.92 ]
Reid 2002	339	76.5 (16.96)	370	76 (15.53)	-	32.3 %	0.50 [ -1.90, 2.90 ]
Small 2000	467	69.69 (18.79)	450	71.2 (18.14)	-	32.6 %	-1.51 [ -3.90, 0.88 ]

-100 -50 0 50 100 Favours control Favours treatment

(Continued . . . )

Psychosocial and psychological interventions for preventing postpartum depression (Review) Copyright © 2008 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. 46

Study or subgroup	Treatment	M(CD)	Control	Maran (CD)	Mean Difference	Weight	Mean Difference
	N	Mean(SD)	N	Mean(SD)	IV,Fixed,95% CI	100.0.0/	IV,Fixed,95% C
Subtotal (95% CI) Heterogeneity: Chi ² = 1.85	1292	0 40) 12 -0.0%	1290			100.0 %	-0.85 [ -2.21, 0.52
Test for overall effect: $Z =$		,					
5 Role functioning (emotior	`	,					
Gunn 1998	232	76.2 (36.1)	243	74.3 (38.5)	+	15.2 %	1.90 [ -4.81, 8.61
Morrell 2000	257	82.4 (31.7)	228	79.5 (35.5)	-	18.9 %	2.90 [ -3.12, 8.92
Reid 2002	339	86.1 (29.52)	370	86.3 (29.82)	•	35.9 %	-0.20 [ -4.57, 4.17
Small 2000	467	73.32 (38.12)	450	78.98 (35.73)	-	30.0 %	-5.66 [ -10.44, -0.88
Subtotal (95% CI)	1295		1291		•	100.0 %	-0.93 [ -3.55, 1.69
Heterogeneity: $Chi^2 = 6.11$	, df = 3 (P =	0.11); I ² =51%					
Test for overall effect: Z = 0 6 Vitality	).70 (P = 0.49	))					
Gunn 1998	232	53.5 (20.1)	243	53.1 (22.3)	+	18.3 %	0.40 [ -3.41, 4.21
Morrell 2000	252	56.1 (21.1)	228	54.7 (21.3)	•	18.5 %	1.40 [ -2.40, 5.20
Reid 2002	339	60.9 (19.84)	370	58.6 (20.2)	-	30.6 %	2.30 [ -0.65, 5.25
Small 2000	467	50.08 (22.37)	450	51.28 (21.79)	+	32.6 %	-1.20 [ -4.06, 1.66
Subtotal (95% CI)	1290		1291		,	100.0 %	0.64 [ -0.99, 2.28
Heterogeneity: $Chi^2 = 2.98$	, df = 3 (P =	0.40); l ² =0.0%					• ·
Test for overall effect: $Z = 0$	).77 (P = 0.44	1)					
7 Social functioning Gunn 1998	232	70 2 (24)	243	794 (219)	-	16.7 %	
		78.3 (24)		79.4 (21.9)	1		-1.10 [ -5.24, 3.04
Morrell 2000	257	83.6 (22)	233	84.1 (23.6)	I	17.5 %	-0.50 [ -4.55, 3.55
Reid 2002	339	88.4 (19.45)	370	87.9 (18.76)	•	36.1 %	0.50 [ -2.32, 3.32
Small 2000	467	78.78 (24.28)	450	80.47 (23.69)	•	29.7 %	-1.69 [ -4.79, 1.41
Subtotal (95% CI)	1295		1296			100.0 %	-0.59 [ -2.29, 1.10
Heterogeneity: $Chi^2 = 1.12$	,	,					
Test for overall effect: Z = ( 8 General health	J.69 (P – 0.45	)					
Gunn 1998	232	74.4 (19.7)	243	74.6 (19)	+	18.2 %	-0.20 [ -3.68, 3.28
Morrell 2000	255	76 (19.4)	230	76.9 (20.4)	+	17.5 %	-0.90 [ -4.45, 2.65
Reid 2002	339	80.4 (17.42)	370	79.5 (17.02)	+	34.3 %	0.90 [ -1.64, 3.44
Small 2000	467	72.2 (20.91)	450	73.22 (21)	-	30.0 %	-1.02 [ -3.73, 1.69
Subtotal (95% CI)	1293		1293			100.0 %	-0.19 [ -1.68, 1.29
Heterogeneity: $Chi^2 = 1.22$		0.75); l ² =0.0%	1275			100.0 /0	-0.17 [ -1.00, 1.27
Test for overall effect: $Z = 0$							
Test for subgroup difference	es: Chi ² = 3.6	5, df = 7 (P = 0.8	2), I ² =0.0%				

#### Analysis 1.12. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 12 Perceived social support at 8, 16, 24 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 12 Perceived social support at 8, 16, 24 weeks postpartum

Study or subgroup	Treatment N	Mean(SD)	Control N	Mean(SD)	Std. Mean Difference IV,Fixed,95% Cl	Weight	Std. Mean Difference IV,Fixed,95% Cl
I Perceived social support	at 0 to 8 weeks	;					
Morrell 2000	260	I 6.7 (6.7)	253	16.6 (7.4)	-	100.0 %	0.01 [ -0.16, 0.19 ]
Subtotal (95% CI)	260		253		•	100.0 %	0.01 [ -0.16, 0.19 ]
Heterogeneity: not applica	ble						
Test for overall effect: $Z =$	0.16 (P = 0.87)						
2 Perceived social support							
Reid 2002	344	5.3 (0.76)	388	5.3 (0.82)		100.0 %	0.0 [ -0.15, 0.15 ]
Subtotal (95% CI)	344		388		+	100.0 %	0.0 [ -0.15, 0.15 ]
Heterogeneity: not applica	ble						
Test for overall effect: $Z =$	, ,						
3 Perceived social support						22.4.44	
Morrell 2000	240	17.1 (6.8)	225	16.7 (7.3)	<b>–</b>	39.6 %	0.06 [ -0.13, 0.24 ]
Reid 2002	339	5.3 (0.66)	370	5.3 (0.71)	<b>•</b>	60.4 %	0.0 [ -0.15, 0.15 ]
Subtotal (95% CI)	579		595		•	100.0 %	0.02 [ -0.09, 0.14 ]
				-4 Favours	-2 0 2 4 s treatment Favours contro	l	

#### Analysis 1.13. Comparison I All interventions versus usual care - various study outcomes, Outcome 13 Breastfeeding duration at 8, 16, 24 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 13 Breastfeeding duration at 8, 16, 24 weeks postpartum

Risk Rati	Weight	Risk Ratio	Control	Treatment	Study or subgroup
M-H,Fixed,95% (		M-H,Fixed,95% Cl	n/N	n/N	
				;	Breastfeeding at 0 to 8 weeks
0.95 [ 0.77, 1.18	33.6 %	+	59/88	55/86	Armstrong 1999
1.07 [ 0.88, 1.29	66.4 %	=	113/268	126/280	Morrell 2000
1.03 [ 0.89, 1.19	100.0 %	+	356	366	Subtotal (95% CI)
				172 (Control)	Total events: 181 (Treatment),
			.0%	= I (P = 0.43); I ² =0.	Heterogeneity: Chi ² = 0.61, df
				(P = 0.70)	Test for overall effect: Z = 0.38
				<s< td=""><td>2 Breastfeeding at 9 to 16 wee</td></s<>	2 Breastfeeding at 9 to 16 wee
0.74 [ 0.52, 1.04	24.9 %		42/80	31/80	Armstrong 1999
0.94 [ 0.79, 1.12	75.1 %	=	130/243	117/232	Gunn 1998
0.89 [ 0.76, 1.04	100.0 %	•	323	312	Subtotal (95% CI)
				172 (Control)	Total events: 148 (Treatment),
			5%	$=   (P = 0.2  );  ^2 = 35$	Heterogeneity: Chi ² = 1.55, df
				(P = 0.15)	Test for overall effect: $Z = 1.45$
				eks	3 Breastfeeding at 17 to 24 we
0.87 [ 0.69, 1.09	65.4 %	-	98/243	81/232	Gunn 1998
0.97 [ 0.68, 1.38	34.6 %	+	48/233	52/260	Morrell 2000
0.90 [ 0.74, 1.10	100.0 %	•	476	492	Subtotal (95% CI)
				146 (Control)	Total events: 133 (Treatment),
			.0%	=   (P = 0.59);   ² = 0.	Heterogeneity: Chi ² = 0.29, df
				(P = 0.30)	Test for overall effect: Z = 1.04

0.1 0.2 0.5 1 2 5 10

Favours control Favours treatment

#### Analysis 1.14. Comparison I All interventions versus usual care - various study outcomes, Outcome 14 Infant immunisations.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 14 Infant immunisations

Study or subgroup	Treatment N	Mean(SD)	Control N	Mean(SD)	Mean Diff IV,Fixed,95%		Mean Difference IV,Fixed,95% CI
I Mean number of imr	nunizations at 12	to 16 weeks					
Armstrong 1999	80	4.92 (0.6)	80	4.5 (1.3)	+	100.0 %	0.42 [ 0.11, 0.73 ]
Total (95% CI)	80		80		*	100.0 %	0.42 [ 0.11, 0.73 ]
Heterogeneity: not app	olicable						
Test for overall effect: 2	Z = 2.62 (P = 0.0)	0087)					
Test for subgroup diffe	rences: Not appli	cable					
					• •	<b>I</b> I	
					-10 -5 0	5 10	
					Favours control Fa	vours treatment	

#### Analysis 1.15. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 15 Infant injuries.

Review: Psychosocial an	d psychological interve	ntions for preventing p	postpartum depres	sion		
Comparison: I All inter	ventions versus usual c	are - various study out	tcomes			
Outcome: 15 Infant inju	ries					
Study or subgroup	Treatment n/N	Control n/N		Risk Ratio red,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Infant injuries at 9 to 16	weeks					
Armstrong 1999	I 5/80	28/80	- <mark></mark>		100.0 %	0.54 [ 0.31, 0.92 ]
Total (95% CI)	80	80	•		100.0 %	0.54 [ 0.31, 0.92 ]
Total events: 15 (Treatmen	, , ,					
Heterogeneity: not applica Test for overall effect: Z =						
	2.21 (1 0.023)					
			0.1 0.2 0.5	1 2 5 10		
			Favours treatment	Favours control		

#### Analysis 1.16. Comparison 1 All interventions versus usual care - various study outcomes, Outcome 16 Marital discord at 4 and 24 weeks postpartum.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: I All interventions versus usual care - various study outcomes

Outcome: 16 Marital discord at 4 and 24 weeks postpartum

Study or subgroup	Treatment N	Mean(SD)	Control N	Mean(SD)	Mean Differenc IV,Fixed,95% Cl	e Weight	Mean Difference IV,Fixed,95% Cl
l Marital discord at 0 to 8	weeks						
Gorman 2002	16	106.8 (19.9)	15	110 (19.1)		100.0 %	-3.20 [ -16.93, 10.53 ]
Subtotal (95% CI)	16		15		+	100.0 %	-3.20 [ -16.93, 10.53 ]
Heterogeneity: not applica	able						
Test for overall effect: Z =	0.46 (P = 0.65	)					
2 Marital discord at 17 to	24 weeks						
Gorman 2002	13	99.5 (21.1)	16	107.4 (15)		100.0 %	-7.90 [ -21.52, 5.72 ]
Subtotal (95% CI)	13		16		•	100.0 %	-7.90 [ -21.52, 5.72 ]
Heterogeneity: not applica	able						
Test for overall effect: Z =	1.14 (P = 0.26	)					
Test for subgroup differen	ces: $Chi^2 = 0.22$	3, df = 1 (P = 0.	.63), I ² =0.0%	, )			
						I	
				-100	) -50 0 50	100	

Favours treatment Favours control

#### Analysis 2.1. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome I Antenatal and postnatal classes.

Review: Psychosocial ar	nd psychological interve	ntions for preventi	ng postpartum depression		
Comparison: 2 Psychos	ocial interventions vers	us usual care - vari	ations in intervention type		
Outcome: I Antenatal	and postnatal classes				
Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Depressive symptomate	ology at final study asses	sment			
Brugha 2000	15/94	18/96		75.0 %	0.85 [ 0.46, 1.59 ]
Stamp 1995	9/60	6/61		25.0 %	1.53 [ 0.58, 4.02 ]
Total (95% CI)	154	157	-	100.0 %	1.02 [ 0.61, 1.72 ]
Total events: 24 (Treatmer	nt), 24 (Control)				
Heterogeneity: Chi ² = 0.9	9, df = 1 (P = 0.32); $I^2$	=0.0%			
Test for overall effect: Z =	: 0.07 (P = 0.94)				
			0.1 0.2 0.5 1 2 5 10		
			Favours treatment Favours control		

#### Analysis 2.2. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 2 Postpartum professional-based home visits.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 2 Psychosocial interventions versus usual care - variations in intervention type

Outcome: 2 Postpartum professional-based home visits

Study or subgroup	Treatment	Control	F	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fi>	ked,95% Cl		M-H,Fixed,95% Cl
I Depressive symptomato	logy at final study asses	sment				
Armstrong 1999	13/80	18/80		-	10.2 %	0.72 [ 0.38, 1.37 ]
MacArthur 2002	5/80	149/702			89.8 %	0.68 [ 0.54, 0.84 ]
Total (95% CI)	881	782	*		100.0 %	0.68 [ 0.55, 0.84 ]
Total events: 128 (Treatme	ent), 167 (Control)					
Heterogeneity: $Chi^2 = 0.0$	4, df = 1 (P = 0.85); l ²	=0.0%				
Test for overall effect: Z =	3.60 (P = 0.00032)					
			0.1 0.2 0.5	1 2 5 10		
			Favours treatment	Favours control		

#### Analysis 2.3. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 3 Postpartum lay-based home visits.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 2 Psychosocial interventions versus usual care - variations in intervention type

Outcome: 3 Postpartum lay-based home visits

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Depressive symptomate	6, ,		_	100.0 %	
Morrell 2000	48/252	49/229		100.0 %	0.89 [ 0.62, 1.27 ]
Total (95% CI) Total events: 48 (Treatme Heterogeneity: not applic Test for overall effect: Z =	able	229		100.0 %	0.89 [ 0.62, 1.27 ]
			0.1 0.2 0.5 1 2 5 10 Favours treatment Favours control		

#### Analysis 2.4. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 4 Early postpartum follow up.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 2 Psychosocial interventions versus usual care - variations in intervention type

Outcome: 4 Early postpartum follow up

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Fixed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Depressive symptomato	logy at final study asses	sment			
Gunn 1998	27/232	31/243		100.0 %	0.91 [ 0.56, 1.48 ]
<b>Total (95% CI)</b> Total events: 27 (Treatmen Heterogeneity: not applica Test for overall effect: Z =	ble	243	-	100.0 %	0.91 [ 0.56, 1.48 ]
			0.1 0.2 0.5 1 2 5 10 Favours treatment Favours control		

#### Analysis 2.5. Comparison 2 Psychosocial interventions versus usual care - variations in intervention type, Outcome 5 Continuity of care.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 2 Psychosocial interventions versus usual care - variations in intervention type

Outcome: 5 Continuity of care

Study or subgroup	Treatment n/N	Control n/N		Risk Ratio xed,95% Cl	Weight	Risk Ratio M-H,Fixed,95% Cl
I Depressive symptomatolo	ogy at final study assessr	ment				
Waldenstrom 2000	74/464	56/471		<b></b>	100.0 %	1.34 [ 0.97, 1.85 ]
Total (95% CI)	464	471		•	100.0 %	1.34 [ 0.97, 1.85 ]
Total events: 74 (Treatment	), 56 (Control)					
Heterogeneity: not applicab	le					
Test for overall effect: $Z = I$	.78 (P = 0.074)					
			0.1 0.2 0.5	1 2 5 10		
			Favours treatment	Favours control		

## Analysis 3.1. Comparison 3 Psychological Interventions versus usual care - variations in intervention type, Outcome I Psychological debriefing.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 3 Psychological Interventions versus usual care - variations in intervention type

Outcome: I Psychological debriefing

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratio M-H,Random,95% Cl
I Depressive symptomatolog	y at final study assessm	ent - all trials			
Gamble 2003	4/50	17/53		14.4 %	0.25 [ 0.09, 0.69 ]
Lavender 1998	5/58	31/56		16.4 %	0.16 [ 0.07, 0.37 ]
Priest 2003	37/696	42/705	+	22.8 %	0.89 [ 0.58, 1.37 ]
Small 2000	81/467	65/450	-	24.3 %	1.20 [ 0.89, 1.62 ]
Tam 2003	26/261	35/255	-	22.1 %	0.73 [ 0.45, 1.17 ]
Subtotal (95% CI) Total events: 153 (Treatment)	<b>1532</b> I, 190 (Control)	1519	•	100.0 %	0.57 [ 0.31, 1.04 ]
			0.01 0.1 1 10 100 Favours treatment Favours control		(Continued )

(Continued . . . )

							( Continued)
Study or subgroup	Treatment	Control		Risk Ratio		Weight	Risk Ratio
	n/N	n/N	1	1-H,Random,95%	CI		M-H,Random,95% Cl
Heterogeneity: Tau ² = 0.37; C	Chi ² = 26.21, df = 4 (P	= 0.00003); I ² =8	5%				
Test for overall effect: Z = 1.8	5 (P = 0.065)						
2 Depressive symptomatolog	y at final study assessm	ent - sensitivity an	alysis				
Gamble 2003	4/50	17/53				18.5 %	0.25 [ 0.09, 0.69 ]
Lavender 1998	5/58	31/56				21.0 %	0.16 [ 0.07, 0.37 ]
Priest 2003	37/696	42/705		+		29.2 %	0.89 [ 0.58, 1.37 ]
Small 2000	81/467	65/450		-		31.2 %	1.20 [ 0.89, 1.62 ]
Subtotal (95% CI)	1271	1264		•		100.0 %	0.51 [ 0.23, 1.14 ]
Total events: 127 (Treatment)	, 155 (Control)						
Heterogeneity: Tau ² = 0.56; C	Chi ² = 25.82, df = 3 (P	$= 0.0000  \text{I}$ ); $I^2 = 8$	8%				
Test for overall effect: Z = 1.6	5 (P = 0.099)						
					1		
			0.01	0.1 1 10	100		
			Favours trea	tment Favours	control		

#### Analysis 3.2. Comparison 3 Psychological Interventions versus usual care - variations in intervention type, Outcome 2 Interpersonal psychotherapy.

Review: Psychosocial and psychological interventions for preventing postpartum depression Comparison: 3 Psychological Interventions versus usual care - variations in intervention type Outcome: 2 Interpersonal psychotherapy Study or subgroup Treatment Control Risk Ratio Weight Risk Ratio n/N n/N M-H,Random,95% Cl M-H,Random,95% Cl I Depressive symptomatology at final study assessment Gorman 2002 4/17 65.7 % 0.64 [ 0.17, 2.46 ] 3/20 Zlotnick 2001 0/17 6/18 34.3 % 0.08 [ 0.00, 1.34 ] 0.31 [ 0.04, 2.52 ] Total (95% CI) 37 100.0 % 35 Total events: 3 (Treatment), 10 (Control) Heterogeneity: Tau² = 1.25; Chi² = 1.99, df = 1 (P = 0.16); l² = 50% Test for overall effect: Z = 1.09 (P = 0.28) 0.001 0.01 0.1 10 100 1000 Favours treatment Favours control

Psychosocial and psychological interventions for preventing postpartum depression (Review) Copyright © 2008 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd. ~

- 2

# Analysis 4.1. Comparison 4 All interventions versus usual care - variations in intervention mode, Outcome I Individually based interventions.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 4 All interventions versus usual care - variations in intervention mode

Outcome: I Individually based interventions

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratio M-H,Random,95% Cl
I Depressive symptomatology	vat 0 to 8 weeks - all 1	rials			
Armstrong 1999	5/86	18/88		9.3 %	0.28 [ 0.11, 0.73 ]
Gorman 2002	0/20	5/20		1.6 %	0.09 [ 0.01, 1.54 ]
Lavender 1998	5/58	31/56	-	10.3 %	0.16 [ 0.07, 0.37 ]
Morrell 2000	49/276	48/266	-	20.1 %	0.98 [ 0.69, 1.41 ]
Priest 2003	54/809	63/833	-	20.3 %	0.88 [ 0.62, 1.25 ]
Tam 2003	26/261	35/255	-	17.5 %	0.73 [ 0.45, 1.17 ]
Waldenstrom 2000	74/464	56/471	-	20.9 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI)	1974	1989	•	100.0 %	0.64 [ 0.40, 1.01 ]
Total events: 213 (Treatment), Heterogeneity: Tau ² = 0.26; C Test for overall effect: $Z = 1.90$ 2 Depressive symptomatology	$Chi^2 = 31.23$ , df = 6 (P 0 (P = 0.057)	,	%		
Armstrong 1999	5/86	18/88	-	11.3 %	0.28 [ 0.11, 0.73 ]
Gorman 2002	0/20	5/20		1.9 %	0.09 [ 0.01, 1.54 ]
Lavender 1998	5/58	31/56	-	12.5 %	0.16 [ 0.07, 0.37 ]
Morrell 2000	49/276	48/266	+	24.4 %	0.98 [ 0.69, 1.41 ]
Priest 2003	54/809	63/833	-	24.6 %	0.88 [ 0.62, 1.25 ]
Waldenstrom 2000	74/464	56/471	-	25.3 %	1.34 [ 0.97, 1.85 ]
<b>Subtotal (95% CI)</b> Total events: 187 (Treatment), Heterogeneity: Tau ² = 0.33; C	$Chi^2 = 30.54, df = 5 (P$	<b>1734</b> = 0.00001);   ² =84	%	100.0 %	0.60 [ 0.34, 1.05 ]
Test for overall effect: $Z = 1.80$ 3 Depressive symptomatology		trials			
Armstrong 1999	13/80	18/80	-	22.0 %	0.72 [ 0.38, 1.37 ]
Gamble 2003	4/50	17/53		13.2 %	0.25 [ 0.09, 0.69 ]
Gunn 1998	38/232	33/243	+	29.0 %	1.21 [ 0.78, 1.85 ]
MacArthur 2002	115/801	149/702	-	35.8 %	0.68 [ 0.54, 0.84 ]
			0.001 0.01 0.1 10 100 1000 Favours treatment Favours control		

(Continued . . . )

	Treatment	Control	Risk Ratio	Weight	( Continue Risk Ratio
	n/N	n/N	M-H,Random,95% Cl		M-H,Random,95% C
Subtotal (95% CI)	1163	1078	•	100.0 %	0.71 [ 0.45, 1.12
Total events: 170 (Treatment), $2$	· /	- 0.02): 12 -70%			
Heterogeneity: Tau ² = 0.13; Cr Test for overall effect: Z = 1.48	,	- 0.02); 170%			
4 Depressive symptomatology		nsitivity analysis			
Armstrong 1999	13/80	18/80	-	34.2 %	0.72 [ 0.38, 1.37
Gamble 2003	4/50	17/53		20.6 %	0.25 [ 0.09, 0.69
Gunn 1998	38/232	33/243	-	45.1 %	1.21 [ 0.78, 1.85
Subtotal (95% CI)	362	376	•	100.0 %	0.67 [ 0.31, 1.48
Total events: 55 (Treatment), 68	, ,				
Heterogeneity: $Tau^2 = 0.35$ ; Ch	,	= 0.02); l ² =76%			
Test for overall effect: Z = 0.99 5 Depressive symptomatology	, ,				
Gorman 2002	3/20	4/17		6.6 %	0.64 [ 0.17, 2.46
Gunn 1998	27/232	31/243	-	20.4 %	0.91 [ 0.56, 1.48
Morrell 2000	48/252	49/229	-	23.8 %	0.89 [ 0.62, 1.27
Priest 2003	55/777	65/797	-	24.0 %	0.87 [ 0.61, 1.23
Small 2000	81/467	65/450	<b>–</b>	25.2 %	1.20 [ 0.89, 1.62
Subtotal (95% CI)	1748	1736	•	100.0 %	0.98 [ 0.82, 1.17
Test for overall effect: Z = 0.24	(P = 0.81)				
Test for overall effect: Z = 0.24 6 Depressive symptomatology	, ,	ent - all trials			
	, ,	ent - all trials 18/80	+	7.9 %	0.72 [ 0.38, 1.37
6 Depressive symptomatology	at final study assessm			7.9 % 4.8 %	-
6 Depressive symptomatology Armstrong 1999	at final study assessm I 3/80	18/80	 		0.25 [ 0.09, 0.69
6 Depressive symptomatology Armstrong 1999 Gamble 2003	at final study assessm I 3/80 4/50	18/80 17/53	 	4.8 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002	at final study assessm 13/80 4/50 3/20	18/80 17/53 4/17	•  •	4.8 % 3.2 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998	at final study assessm 13/80 4/50 3/20 27/232	18/80 17/53 4/17 31/243	   	4.8 % 3.2 % 9.8 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998	at final study assessm 13/80 4/50 3/20 27/232 5/58	18/80 17/53 4/17 31/243 31/56	   	4.8 % 3.2 % 9.8 % 5.8 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801	18/80 17/53 4/17 31/243 31/56 149/702	+  + +	4.8 % 3.2 % 9.8 % 5.8 % 12.9 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252	18/80 17/53 4/17 31/243 31/56 149/702 49/229		4.8 % 3.2 % 9.8 % 5.8 % 12.9 % 11.4 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000 Priest 2003	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252 37/696	18/80 17/53 4/17 31/243 31/56 149/702 49/229 42/705		4.8 % 3.2 % 9.8 % 5.8 % 12.9 % 11.4 % 10.5 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37 1.20 [ 0.89, 1.62
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000 Priest 2003 Small 2000	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252 37/696 81/467	18/80 17/53 4/17 31/243 31/56 149/702 49/229 42/705 65/450		4.8 % 3.2 % 9.8 % 12.9 % 11.4 % 10.5 % 12.1 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37 1.20 [ 0.89, 1.62 0.73 [ 0.45, 1.17
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000 Priest 2003 Small 2000 Tam 2003 Waldenstrom 2000	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252 37/696 81/467 26/261 74/464 <b>3381</b>	18/80 17/53 4/17 31/243 31/56 149/702 49/229 42/705 65/450 35/255		4.8 % 3.2 % 9.8 % 12.9 % 11.4 % 10.5 % 12.1 % 9.9 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37 1.20 [ 0.89, 1.62 0.73 [ 0.45, 1.17 1.34 [ 0.97, 1.85
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000 Priest 2003 Small 2000 Tam 2003 Waldenstrom 2000 Subtotal (95% CI)	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252 37/696 81/467 26/261 74/464 <b>3381</b> 497 (Control)	18/80 17/53 4/17 31/243 31/56 149/702 49/229 42/705 65/450 35/255 56/471 <b>3261</b>		4.8 % 3.2 % 9.8 % 5.8 % 12.9 % 11.4 % 10.5 % 12.1 % 9.9 % 11.8 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37 1.20 [ 0.89, 1.62 0.73 [ 0.45, 1.17 1.34 [ 0.97, 1.85
6 Depressive symptomatology : Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000 Priest 2003 Small 2000 Tam 2003 Waldenstrom 2000	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252 37/696 81/467 26/261 74/464 <b>3381</b> 497 (Control)	18/80 17/53 4/17 31/243 31/56 149/702 49/229 42/705 65/450 35/255 56/471 <b>3261</b>		4.8 % 3.2 % 9.8 % 5.8 % 12.9 % 11.4 % 10.5 % 12.1 % 9.9 % 11.8 %	0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37 1.20 [ 0.89, 1.62 0.73 [ 0.45, 1.17 1.34 [ 0.97, 1.85
6 Depressive symptomatology Armstrong 1999 Gamble 2003 Gorman 2002 Gunn 1998 Lavender 1998 MacArthur 2002 Morrell 2000 Priest 2003 Small 2000 Tam 2003 Waldenstrom 2000 Subtotal (95% CI)	at final study assessm 13/80 4/50 3/20 27/232 5/58 115/801 48/252 37/696 81/467 26/261 74/464 <b>3381</b> 497 (Control)	18/80 17/53 4/17 31/243 31/56 149/702 49/229 42/705 65/450 35/255 56/471 <b>3261</b> P = 0.00003); l ² =74%	6 0.001 0.01 0.1 10 100 1000 Favours treatment	4.8 % 3.2 % 9.8 % 5.8 % 12.9 % 11.4 % 10.5 % 12.1 % 9.9 % 11.8 %	0.72 [ 0.38, 1.37 0.25 [ 0.09, 0.69 0.64 [ 0.17, 2.46 0.91 [ 0.56, 1.48 0.16 [ 0.07, 0.37 0.68 [ 0.54, 0.84 0.89 [ 0.62, 1.27 0.89 [ 0.58, 1.37 1.20 [ 0.89, 1.62 0.73 [ 0.45, 1.17 1.34 [ 0.97, 1.85 <b>0.76 [ 0.59, 1.00</b>

Psychosocial and psychological interventions for preventing postpartum depression (Review) Copyright © 2008 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

57

	Study or subgroup	Control n/N	Risk Ratio	Weight	( Continued Risk Ratio
	Test for everall offects 7 - 10	n/IN	M-H,Random,95% Cl		M-H,Random,95% Cl
	Test for overall effect: Z = 1.9 7 Depressive symptomatology	ant annaiti itu anab ai			
	. ,		s 	142.0/	
	Armstrong 1999	18/80		14.2 %	0.72 [ 0.38, 1.37 ]
	Gamble 2003	17/53		8.5 %	0.25 [ 0.09, 0.69 ]
	Gorman 2002	4/17		5.7 %	0.64 [ 0.17, 2.46 ]
	Sonnan 2002			517 75	
	Lavender 1998	31/56		10.4 %	0.16 [ 0.07, 0.37 ]
	Priest 2003	42/705	+	18.7 %	0.89 [ 0.58, 1.37 ]
	Small 2000	65/450	+	21.5 %	1.20 [ 0.89, 1.62 ]
	Waldenstrom 2000	56/471	-	21.0 %	1.34 [ 0.97, 1.85 ]
	Subtotal (95% CI)	1832	•	100.0 %	0.68 [ 0.43, 1.09 ]
	Total events: 217 (Treatment),				
= 6 (P = 0	Heterogeneity: $Tau^2 = 0.28$ ; C	= 0.00002); I ² =81%			
	Test for overall effect: $Z = 1.5$				

0.001 0.01 0.1 1 10 100 1000

Favours control

Favours treatment

# Analysis 4.2. Comparison 4 All interventions versus usual care - variations in intervention mode, Outcome 2 Group-based interventions.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 4 All interventions versus usual care - variations in intervention mode

Outcome: 2 Group-based interventions

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratic M-H,Random,95% C
		n/IN	M-H,Random,95% CI		M-H,Kandom,95% C
I Depressive symptomatolog Stamp 1995	y at 0 to 8 weeks 8/64	11/64	<b>—</b>	100.0 %	0.73 [ 0.31, 1.69
Subtotal (95% CI)	64	64	•	100.0 %	0.73 [ 0.31, 1.69 ]
Total events: 8 (Treatment), 1 Heterogeneity: not applicable	. ,				
Test for overall effect: $Z = 0.7$					
2 Depressive symptomatolog	· /	trials			
Brugha 2000	15/94	18/96	-	25.0 %	0.85 [ 0.46, 1.59
Reid 2002	55/344	46/388	=	60.9 %	1.35 [ 0.94, 1.94
Stamp 1995	7/63	10/65		12.7 %	0.72 [ 0.29, 1.78
Zlotnick 2001	0/17	6/18	• • • • • • • • • • • • • • • • • • •	1.4 %	0.08 [ 0.00, 1.34
Subtotal (95% CI)	518	567	•	100.0 %	0.93 [ 0.54, 1.59
Test for overall effect. $Z = 0.2$					
Test for overall effect: Z = 0.2 3 Depressive symptomatolog	y at 9 to 16 weeks - se	, ,			
	· /	nsitivity analysis 18/96	+	25.4 %	0.85 [ 0.46, 1.59
3 Depressive symptomatolog	y at 9 to 16 weeks - se	, ,	-	25.4 % 61.8 %	2
3 Depressive symptomatolog Brugha 2000	y at 9 to 16 weeks - se 15/94	18/96	-		1.35 [ 0.94, 1.94
3 Depressive symptomatolog Brugha 2000 Reid 2002	y at 9 to 16 weeks - se 15/94 55/344	18/96 46/388	-	61.8 %	0.85 [ 0.46, 1.59 1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78 <b>1.08 [ 0.74, 1.57</b> ]
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment),	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control)	18/96 46/388 10/65 <b>549</b>	•	61.8 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P =	18/96 46/388 10/65 <b>549</b>	• •	61.8 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C Test for overall effect: Z = 0.3	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P = 99 (P = 0.69)	18/96 46/388 10/65 <b>549</b>	•	61.8 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P = 99 (P = 0.69)	18/96 46/388 10/65 <b>549</b>	-	61.8 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C Test for overall effect: Z = 0.3 4 Depressive symptomatolog	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P = 99 (P = 0.69) y at 17 to 24 weeks	18/96 46/388 10/65 <b>549</b> = 0.27); I ² =24%	•	61.8 % 12.8 % <b>100.0 %</b>	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78 <b>1.08 [ 0.74, 1.57</b> 1.16 [ 0.80, 1.69
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C Test for overall effect: Z = 0.3 4 Depressive symptomatolog Reid 2002 Stamp 1995	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P = 19 (P = 0.69) y at 17 to 24 weeks 49/339	18/96 46/388 10/65 <b>549</b> = 0.27); I ² =24% 46/370	•	61.8 % 12.8 % <b>100.0 %</b> 84.1 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78 <b>1.08 [ 0.74, 1.57</b> 1.16 [ 0.80, 1.69 1.53 [ 0.58, 4.02
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C Test for overall effect: Z = 0.3 4 Depressive symptomatolog Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b>	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Ch ² = 2.65, df = 2 (P = 19 (P = 0.69) y at 17 to 24 weeks 49/339 9/60 <b>399</b>	18/96 46/388 10/65 <b>549</b> = 0.27); I ² =24% 46/370 6/61	•	61.8 % 12.8 % <b>100.0 %</b> 84.1 % 15.9 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78 <b>1.08 [ 0.74, 1.57</b> 1.16 [ 0.80, 1.69
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C Test for overall effect: Z = 0.3 4 Depressive symptomatolog Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 58 (Treatment),	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P = 19 (P = 0.69) y at 17 to 24 weeks 49/339 9/60 <b>399</b> 52 (Control)	18/96 46/388 10/65 <b>549</b> = 0.27); 1 ² =24% 46/370 6/61 <b>431</b>	•	61.8 % 12.8 % <b>100.0 %</b> 84.1 % 15.9 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78 <b>1.08 [ 0.74, 1.57</b> 1.16 [ 0.80, 1.69 1.53 [ 0.58, 4.02
3 Depressive symptomatolog Brugha 2000 Reid 2002 Stamp 1995 <b>Subtotal (95% CI)</b> Total events: 77 (Treatment), Heterogeneity: Tau ² = 0.03; C Test for overall effect: Z = 0.3 4 Depressive symptomatolog Reid 2002	y at 9 to 16 weeks - se 15/94 55/344 7/63 <b>501</b> 74 (Control) Chi ² = 2.65, df = 2 (P = 19 (P = 0.69) y at 17 to 24 weeks 49/339 9/60 <b>399</b> 52 (Control) ni ² = 0.26, df = 1 (P =	18/96 46/388 10/65 <b>549</b> = 0.27); 1 ² =24% 46/370 6/61 <b>431</b>		61.8 % 12.8 % <b>100.0 %</b> 84.1 % 15.9 %	1.35 [ 0.94, 1.94 0.72 [ 0.29, 1.78 <b>1.08 [ 0.74, 1.57</b> 1.16 [ 0.80, 1.69 1.53 [ 0.58, 4.02

Favours treatment Favours control

(Continued . . . )

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% CI	Weight	( Continued) Risk Ratio M-H,Random,95% Cl
Brugha 2000	15/94	18/96	-	26.1 %	0.85 [ 0.46, 1.59 ]
Reid 2002	49/339	46/370	•	60.9 %	1.16 [ 0.80, 1.69 ]
Stamp 1995	9/60	6/61		11.5 %	1.53 [ 0.58, 4.02 ]
Zlotnick 2001	0/17	6/18	·	1.4 %	0.08 [ 0.00, 1.34 ]
Subtotal (95% CI)	510	545	•	100.0 %	1.03 [ 0.65, 1.63 ]
Total events: 73 (Treatment), Heterogeneity: $Tau^2 = 0.07$ ; C Test for overall effect: $Z = 0.1$	Chi ² = 4.57, df = 3 (P =	= 0.21); I ² =34%			
			0.001 0.01 0.1 10 100 1000		

Favours treatment Favours control

#### Analysis 5.1. Comparison 5 All interventions versus usual care - variations in intervention onset, Outcome I Interventions with antenatal and postnatal components.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 5 All interventions versus usual care - variations in intervention onset

Outcome: I Interventions with antenatal and postnatal components

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% Cl	, reight	M-H,Fixed,95% Cl
I Depressive symptomatolo	egy at final study assess	ment			
Brugha 2000	15/94	18/96		21.3 %	0.85 [ 0.46, 1.59 ]
Gorman 2002	3/20	4/17		5.2 %	0.64 [ 0.17, 2.46 ]
Stamp 1995	9/60	6/61		7.1 %	1.53 [ 0.58, 4.02 ]
Waldenstrom 2000	74/464	56/471	-	66.4 %	1.34 [ 0.97, 1.85 ]
Total (95% CI)	638	645	•	100.0 %	1.21 [ 0.93, 1.59 ]
Total events: 101 (Treatmen	t), 84 (Control)				
Heterogeneity: Chi ² = 2.70,	df = 3 (P = 0.44); $I^2$ =	=0.0%			
Test for overall effect: $Z = I$	.42 (P = 0.16)				
			0.1 0.2 0.5 1 2 5 10		

Favours treatment Favours control

Psychosocial and psychological interventions for preventing postpartum depression (Review) Copyright © 2008 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

# Analysis 5.2. Comparison 5 All interventions versus usual care - variations in intervention onset, Outcome 2 Interventions with postnatal only component.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 5 All interventions versus usual care - variations in intervention onset

Outcome: 2 Interventions with postnatal only component

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Random,95% Cl		M-H,Random,95% CI
I Depressive symptomatology	at final study assessm	ent - all trials			
Armstrong 1999	13/80	18/80		8.4 %	0.72 [ 0.38, 1.37 ]
Gamble 2003	4/50	17/53		5.2 %	0.25 [ 0.09, 0.69 ]
Gunn 1998	27/232	31/243	-	10.2 %	0.91 [ 0.56, 1.48 ]
Lavender 1998	5/58	31/56		6.2 %	0.16 [ 0.07, 0.37 ]
MacArthur 2002	115/801	149/702	-	13.2 %	0.68 [ 0.54, 0.84 ]
Morrell 2000	48/252	49/229	+	11.8 %	0.89 [ 0.62,  .27 ]
Priest 2003	37/696	42/705	-	10.9 %	0.89 [ 0.58, 1.37 ]
Reid 2002	49/339	46/370	+	11.5 %	1.16 [ 0.80, 1.69 ]
Small 2000	81/467	65/450	-	12.4 %	1.20 [ 0.89, 1.62 ]
Tam 2003	26/261	35/255		10.3 %	0.73 [ 0.45, 1.17 ]
Subtotal (95% CI)	3236	3143	•	100.0 %	0.76 [ 0.58, 0.98 ]
Heterogeneity: Tau ² = 0.11; Cl Test for overall effect: Z = 2.08 2 Depressive symptomatology	B (P = 0.037)	,	s		
Armstrong 1999	13/80	18/80		19.5 %	0.72 [ 0.38, 1.37 ]
Gamble 2003	4/50	17/53		12.0 %	0.25 [ 0.09, 0.69 ]
Lavender 1998	5/58	31/56		14.5 %	0.16 [ 0.07, 0.37 ]
Priest 2003	37/696	42/705	-	25.3 %	0.89 [ 0.58, 1.37 ]
Small 2000	81/467	65/450	-	28.8 %	1.20 [ 0.89, 1.62 ]
Subtotal (95% CI)	1351	1344	•	100.0 %	0.56 [ 0.29, 1.06 ]
Total events: 140 (Treatment),	. ,	= 0.00003); l ² =85%			

#### Analysis 6.1. Comparison 6 All interventions versus usual care - variations in intervention duration, Outcome I Single-contact interventions.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 6 All interventions versus usual care - variations in intervention duration

Outcome: I Single-contact interventions

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratic M-H,Random,95% C
		n/IN	M-H,Kandom,95% CI		I*I-H,Kandom,95% C
I Depressive symptomatology Lavender 1998	at 0 to 8 weeks 5/58	31/56		35.8 %	0.16 [ 0.07, 0.37
Priest 2003	54/809	63/833	- T	64.2 %	0.88 [ 0.62, 1.25 ]
Subtotal (95% CI)	867	889		100.0 %	0.39 [ 0.07, 2.16 ]
Total events: 59 (Treatment), 9	· ,	2			
Heterogeneity: $Tau^2 = 1.42$ ; Cl	,	$= 0.00025); 1^2 = 93\%$			
Test for overall effect: Z = 1.08 2 Depressive symptomatology	, ,				
Gunn 1998	38/232	33/243		100.0 %	1.21 [ 0.78, 1.85
Subtotal (95% CI)	232	243	•	100.0 %	1.21 [ 0.78, 1.85
Total events: 38 (Treatment), 3		215		100.0 /0	1.21 [ 0.7 0, 1.09
Heterogeneity: not applicable					
Test for overall effect: $Z = 0.85$	(P = 0.39)				
3 Depressive symptomatology	at 17 to 24 weeks				
Gunn 1998	27/232	31/243	+	30.0 %	0.91 [ 0.56, 1.48
Priest 2003	55/777	65/797	-	34.3 %	0.87 [ 0.61, 1.23
Small 2000	81/467	65/450	-	35.7 %	1.20 [ 0.89, 1.62
Subtotal (95% CI)	1476	1490	•	100.0 %	1.02 [ 0.82, 1.26
Total events: 163 (Treatment), Heterogeneity: Tau ² = 0.00; Cł	. ,	= 0.33);   ² =9%			
Test for overall effect: $Z = 0.14$		,			
4 Depressive symptomatology	at final study assessm	ent - all trials			
Gunn 1998	27/232	31/234	-	25.8 %	0.88 [ 0.54, 1.42
Lavender 1998	5/58	31/56		16.3 %	0.16 [ 0.07, 0.37
Priest 2003	37/696	42/705	+	27.3 %	0.89 [ 0.58, 1.37
Small 2000	81/467	65/450	+	30.6 %	1.20 [ 0.89, 1.62
Subtotal (95% CI)	1453	1445	•	100.0 %	0.70 [ 0.38, 1.27
Total events: 150 (Treatment),					
Heterogeneity: $Tau^2 = 0.30$ ; Ch	,	= 0.00022); I ² =85%			
Test for overall effect: $Z = 1.18$	, ,				
5 Depressive symptomatology	at final study assessm	ent - sensitivity analys	s		
			0.01 0.1 10 100		
		E	avours treatment Favours control		

(Continued . . . )

Study or subgroup	Treatment n/N	Control n/N			Risk Ratio 1dom,95% C		Weight	( Continued) Risk Ratio M-H,Random,95% Cl
Lavender 1998	5/58	31/56		-			22.0 %	0.16 [ 0.07, 0.37 ]
Priest 2003	37/696	42/705		+	-		36.7 %	0.89 [ 0.58, 1.37 ]
Small 2000	81/467	65/450			-		41.3 %	1.20 [ 0.89, 1.62 ]
Subtotal (95% CI)	1221	1211		-	•		100.0 %	0.61 [ 0.26, 1.45 ]
Total events: 123 (Treatment)	, 138 (Control)							
Heterogeneity: Tau ² = 0.49; (	Chi ² = 19.44, df = 2 (P	= 0.00006); l ² =90%	6					
Test for overall effect: $Z = 1.1$	I (P = 0.26)							
			0.01	0.1	1 10	100		
		I	Favours t	reatment	Favours	control		

#### Analysis 6.2. Comparison 6 All interventions versus usual care - variations in intervention duration, **Outcome 2 Multiple-contact interventions.**

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 6 All interventions versus usual care - variations in intervention duration

Outcome: 2 Multiple-contact interventions

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Random,95% CI		M-H,Random,95% Cl
I Depressive symptomatology	v at 0 to 8 weeks - all 1	trials			
Armstrong 1999	5/86	18/88		9.9 %	0.28 [ 0.11, 0.73 ]
Gorman 2002	0/20	5/20		1.4 %	0.09 [ 0.01, 1.54 ]
Morrell 2000	48/252	49/229	•	26.9 %	0.89 [ 0.62, 1.27 ]
Stamp 1995	8/64	11/64	-	11.6 %	0.73 [ 0.31, 1.69 ]
Tam 2003	26/261	35/255	-	21.9 %	0.73 [ 0.45, 1.17 ]
Waldenstrom 2000	74/464	56/471	•	28.3 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI)	1147	1127	•	100.0 %	0.77 [ 0.50, 1.17 ]
Total events: 161 (Treatment),	174 (Control)				
Heterogeneity: Tau ² = 0.15; C	chi ² = 15.12, df = 5 (P	$= 0.01$ ); $ ^2 = 67\%$			
Test for overall effect: $Z = 1.24$	4 (P = 0.21)				
2 Depressive symptomatology	vat 0 to 8 weeks - sen	sitivity analysis			
Armstrong 1999	5/86	18/88	-	12.6 %	0.28 [ 0.11, 0.73 ]
			0.001 0.01 0.1 10 100 1000		
			Favours treatment Favours control		(- )

(Continued . . . )

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	( Continued) Risk Ratio M-H,Random,95% Cl
Gorman 2002	0/20	5/20		1.8 %	0.09 [ 0.01, 1.54 ]
Morrell 2000	48/252	49/229	-	34.4 %	0.89 [ 0.62, 1.27 ]
Stamp 1995	8/64	11/64		14.9 %	0.73 [ 0.31, 1.69 ]
Waldenstrom 2000	74/464	56/471	-	36.2 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI)	886	872	•	100.0 %	0.75 [ 0.44, 1.27 ]
Total events: 135 (Treatment), Heterogeneity: Tau ² = 0.21; C Test for overall effect: $Z = 1.07$ 3 Depressive symptomatology	$hi^2 = 13.82, df = 4 (P)$ 7 (P = 0.29)	,			
Armstrong 1999	13/80	18/80	-	14.5 %	0.72 [ 0.38, 1.37 ]
Brugha 2000	15/94	18/96	+	15.0 %	0.85 [ 0.46, 1.59 ]
Gamble 2003	4/50	17/53		7.8 %	0.25 [ 0.09, 0.69 ]
MacArthur 2002	5/80	149/702	-	28.9 %	0.68 [ 0.54, 0.84 ]
Reid 2002	49/339	46/370	-	23.1 %	1.16 [ 0.80, 1.69 ]
Stamp 1995	7/63	10/65		9.4 %	0.72 [ 0.29, 1.78 ]
Zlotnick 2001	0/17	6/18	•	1.3 %	0.08 [ 0.00, 1.34 ]
Subtotal (95% CI) Total events: 203 (Treatment), Heterogeneity: Tau ² = 0.09; Cl Test for overall effect: Z = 1.91 4 Depressive symptomatology Armstrong 1999	hi ² = 13.24, df = 6 (P 1 (P = 0.056)	,		<b>100.0 %</b> 20.8 %	0.72 [ 0.52, 1.01 ] 0.72 [ 0.38, 1.37 ]
Brugha 2000	15/94	18/96		21.5 %	0.85 [ 0.46, 1.59 ]
Gamble 2003	4/50	17/53		11.2 %	0.25 [ 0.09, 0.69 ]
Reid 2002	49/339	46/370	_	33.1 %	1.16 [ 0.80, 1.69 ]
Stamp 1995	7/63	10/65		13.4 %	
					0.72 [ 0.29, 1.78 ]
Subtotal (95% CI) Total events: 88 (Treatment), 1 Heterogeneity: Tau ² = 0.12; C Test for overall effect: Z = 1.25 5 Depressive symptomatology Gorman 2002	$hi^2 = 8.62, df = 4 (P = 5 (P = 0.21))$	<b>664</b> = 0.07); I ² =54% 4/17	_	<b>100.0 %</b> 8.2 %	0.76 [ 0.49, 1.17 ]
Morrell 2000	48/252	49/229	-	39.5 %	0.89 [ 0.62, 1.27 ]
Reid 2002	49/339	46/370	-	38.3 %	1.16 [ 0.80, 1.69 ]
		6/61	-	14.0 %	1.53 [ 0.58, 4.02 ]
Stamp 1995	9/60	0/01			

0.001 0.01 0.1 1 10 100 1000

Favours treatment Favours control

⁽Continued . . . )

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	( Continuer Risk Ratic M-H,Random,95% Cl
Total events: 109 (Treatment), Heterogeneity: Tau ² = 0.0; Ch Test for overall effect: Z = 0.17	i ² = 2.16, df = 3 (P =	0.54); I ² =0.0%			
6 Depressive symptomatology	at final study assessm	ent - all trials			
Armstrong 1999	13/80	18/80		8.4 %	0.72 [ 0.38, 1.37 ]
Brugha 2000	15/94	18/96	+	8.7 %	0.85 [ 0.46, 1.59 ]
Gamble 2003	4/50	17/53		4.5 %	0.25 [ 0.09, 0.69
Gorman 2002	3/20	4/17		2.9 %	0.64 [ 0.17, 2.46
MacArthur 2002	5/80	149/702	-	16.8 %	0.68 [ 0.54, 0.84 ]
Morrell 2000	48/252	49/229	+	13.8 %	0.89 [ 0.62, 1.27
Reid 2002	49/339	46/370	+	13.4 %	1.16 [ 0.80, 1.69
Stamp 1995	9/60	6/61		4.9 %	1.53 [ 0.58, 4.02
Tam 2003	26/261	35/255	-	11.3 %	0.73 [ 0.45, 1.17
Waldenstrom 2000	74/464	56/471	-	14.6 %	1.34 [ 0.97, 1.85
Zlotnick 2001	0/17	6/18	•	0.8 %	0.08 [ 0.00, 1.34
Subtotal (95% CI)	2438	2352	•	100.0 %	0.84 [ 0.66, 1.08
Total events: 356 (Treatment), Heterogeneity: Tau ² = 0.09; C Test for overall effect: Z = 1.34 7 Depressive symptomatology	$hi^2 = 25.06, df = 10$ ( 4 (P = 0.18) at final study assessm	ent - sensitivity analysis	;		
Armstrong 1999	13/80	18/80		19.1 %	0.72 [ 0.38, 1.37
Brugha 2000	15/94	18/96	-	19.8 %	0.85 [ 0.46, 1.59 ]
Gamble 2003	4/50	17/53		10.3 %	0.25 [ 0.09, 0.69
Gorman 2002	3/20	4/17		6.5 %	0.64 [ 0.17, 2.46
Stamp 1995	9/60	6/61		11.1 %	1.53 [ 0.58, 4.02
Waldenstrom 2000	74/464	56/471	-	33.1 %	1.34 [ 0.97, 1.85
Subtotal (95% CI) Total events: 118 (Treatment), Heterogeneity: Tau ² = 0.18; C Test for overall effect: Z = 0.72	$hi^2 = 12.71, df = 5 (P$	<b>778</b> = 0.03);   ² = 6   %	•	100.0 %	0.84 [ 0.53, 1.34
		F	0.001 0.01 0.1 10 100 1000 avours treatment Favours control		

#### Analysis 7.1. Comparison 7 All interventions versus usual care - variations in risk status, Outcome I Interventions for at-risk women.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 7 All interventions versus usual care - variations in risk status

Outcome: I Interventions for at-risk women

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	M-H,Fixed,95% CI		M-H,Fixed,95% Cl
Depressive symptomatolo	ogy at final study asses	sment			
Armstrong 1999	13/80	18/80	-	17.3 %	0.72 [ 0.38, 1.37 ]
Brugha 2000	15/94	18/96	+	17.1 %	0.85 [ 0.46, 1.59 ]
Gamble 2003	4/50	17/53		15.8 %	0.25 [ 0.09, 0.69 ]
Gorman 2002	3/20	4/17		4.1 %	0.64 [ 0.17, 2.46 ]
Stamp 1995	9/60	6/61		5.7 %	1.53 [ 0.58, 4.02 ]
Tam 2003	26/261	35/255	-	33.9 %	0.73 [ 0.45, 1.17 ]
Zlotnick 2001	0/17	6/18	·	6.1 %	0.08 [ 0.00, 1.34 ]
<b>Fotal (95% CI)</b> otal events: 70 (Treatment leterogeneity: Chi ² = 9.26,	, , ,	580	•	100.0 %	0.67 [ 0.51, 0.89 ]
	$h, dt = 6 (P = 0.16); 1^{2}$	=35%			

Favours treatment Favours control

### Analysis 7.2. Comparison 7 All interventions versus usual care - variations in risk status, Outcome 2 Interventions for general population.

Review: Psychosocial and psychological interventions for preventing postpartum depression

Comparison: 7 All interventions versus usual care - variations in risk status

Outcome: 2 Interventions for general population

Study or subgroup	Treatment n/N	Control n/N	Risk Ratio M-H,Random,95% Cl	Weight	Risk Ratio M-H,Random,95% Cl
I Depressive symptomatology	y at final study assessm	ent - all trials			
Gunn 1998	27/232	31/243	-	11.5 %	0.91 [ 0.56, 1.48 ]
Lavender 1998	5/58	31/56	-	6.9 %	0.16 [ 0.07, 0.37 ]
MacArthur 2002	5/80	149/702	-	15.0 %	0.68 [ 0.54, 0.84 ]
Morrell 2000	48/252	49/229	+	13.3 %	0.89 [ 0.62, 1.27 ]
Priest 2003	37/696	42/705	+	12.3 %	0.89 [ 0.58, 1.37 ]
Reid 2002	49/339	46/370	+	13.1 %	1.16 [ 0.80, 1.69 ]
Small 2000	81/467	65/450	+	14.1 %	1.20 [ 0.89, 1.62 ]
Waldenstrom 2000	74/464	56/471	+	13.8 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI) Total events: 436 (Treatment). Heterogeneity: Tau ² = 0.12; C Test for overall effect: $Z = 0.9$	$Chi^2 = 33.3 I, df = 7 (P)$ 3 (P = 0.35)	,		100.0 %	0.87 [ 0.66, 1.16 ]
2 Depressive symptomatology Lavender 1998	y at final study assessm 5/58	ent - sensitivity a 31/56	nalysis —	14.7 %	0.16 [ 0.07, 0.37 ]
Priest 2003	37/696	42/705	-	26.1 %	0.89 [ 0.58, 1.37 ]
Small 2000	81/467	65/450	<b>–</b>	29.9 %	1.20 [ 0.89, 1.62 ]
Waldenstrom 2000	74/464	56/471		29.3 %	1.34 [ 0.97, 1.85 ]
Subtotal (95% CI)	1685	1682	•	100.0 %	0.80 [ 0.45, 1.40 ]
Total events: 197 (Treatment). Heterogeneity: Tau ² = 0.27; C Test for overall effect: $Z = 0.7$	Chi ² = 22.35, df = 3 (P	= 0.00006); I ² =			
			0.01 0.1 10 100		
			Favours treatment Favours control		

## WHAT'S NEW

Last assessed as up-to-date: 15 August 2004.

Date	Event	Description
8 May 2008	Amended	Converted to new review format.

## HISTORY

Protocol first published: Issue 1, 2001

Review first published: Issue 4, 2004

#### CONTRIBUTIONS OF AUTHORS

Cindy-Lee Dennis updated the previously published protocol and searched for relevant studies with input from Josephine Kavanagh. She also independently evaluated the trials for quality, extracted and entered data, completed the meta-analysis, and wrote the text of the Review and the conclusion. Debra Creedy independently evaluated the trials for quality and extracted and entered data.

## DECLARATIONS OF INTEREST

Dr Dennis is a principal investigator for a multi-site trial, currently on-going, that is evaluating the effect of telephone-based peer (mother-to-mother) support in the prevention of postpartum depression among mothers identified as high-risk. Dr Creedy is a co-investigator on one trial included in this Review.

## SOURCES OF SUPPORT

#### Internal sources

• University of Toronto, Canada.

#### **External sources**

• No sources of support supplied

## ΝΟΤΕS

The title of the previously published protocol was 'Psychosocial interventions for preventing postpartum depression'.

## INDEX TERMS

#### Medical Subject Headings (MeSH)

Depression, Postpartum [* prevention & control]; Family Health; Psychotherapy; Randomized Controlled Trials as Topic; Social Support

### MeSH check words

Female; Humans